Content

Section 1 Introduction

1.1	How to Use This Manual1-	-1
1.2	Optional Accessories1	-2

Section 2 Specifications and System Planning2-1

2.1	Electrical Specifications	. 2-1
2.2	Environmental Specifications	. 2-1
2.3	Wiring Specifications	. 2-2

3.1	1 Federal Communications Commission (FCC)		
	3.1.1 FCC Warning		
3.2	2 Underwriters Laboratories (UL)		
	3.2.1 Requirements for All Installations		
	3.2.2 Requirements for Central Station Fire Alarm Sy	vstems	
	3.2.3 Requirements for Local Protected Fire Alarm S	ystems	3-3
	3.2.4 Requirements for Auxiliary Protected Fire Alar	m Systems for Fire Alarm Service	3-3
	3.2.5 Requirements for Remote Station Protected Fire	e Alarm Systems - Polarity Reversal	3-3
3.3	.3 California Fire Marshal (CFM)		
3.4	4 Factory Mutual (FM)		
3.5	.5 Materials & Equipment Board of Acceptance (MEA)	

Section 4 Installation Overview _____4-1

4.1 Model 5204 Wiring Diagram	
4.2 Current Draw Worksheet	
4.2.1 Worksheet Example	
4.2.2 Worksheet Requirements	4-A

5.1	Grounding the Model 5204 Cover	5-2
5.2	Smoke Power Selection	. 5-2
5.3	Power Supply Wiring	. 5-2
5.4	Battery Connection	. 5-3

5.5	Mounting the 5204	
5.6	Terminal Strip Description	
5.7	Model 5205 Dialer and Telephone Line Connection (Optional)	
	Installation	
	Ring Detect Circuit	
5.8	Cable Connectors	
	Status (P1)	
	Model 5230 (P2)	
	Power Supply (AC) Connector (P4)	

6.1 Zone Wiring	6-1
6.1.1 Four-Wire Smoke Detector Connection	
6.1.2 Two-Wire Smoke Detector Connection	6-3
Notes for Both Tables	6-3
6.2 Connections to Compatible Silent Knight Products	6-6
6.2.1 Model 4180 Status Display Module	6-6
6.2.2 Model 5220 Direct Connect Module	6-7
6.2.2.1 Installation	6-8
6.2.2.2 City Box Connect (24 VDC Systems Only)	6-8
6.2.2.3 NFPA 72 Polarity Reversal (12 or 24 VDC Systems)	6-9
6.2.3 Model 5230 Remote Annunciator	6-9
6.2.3.1 Setting ID Codes	6-9
6.2.3.2 Wiring the 5230 Remote Annunciator	6-10
6.2.3.3 Mounting the 5230 Remote Annunciator	6-11
6.2.4 Model 5295 Signal Power Expander	6-11
6.2.5 Model 7181 Zone Converter	6-12
6.3 Supervised Notification Device Outputs	6-13
6.4 Auxiliary Relays	6-15
6.5 External Silence Keyswitch (Optional)	6-16

Section 7 Normal Operation 7-1

7.1 Buil	It-in Touchpad and Model 5230 Annunciator Operation	
7.1.1	Operating Modes	
7.1.2	Built-in Touchpad Display Codes	
7.1.3	Silencing the System	
7.1.4	LED Indicators	
7.2 Syst	tem Testing	
7.2.1	Fire Drills (Mode 20)	
7.2.2	Walk Test (Mode 22)	
7.2.3	Automatic Self Test	
7.2.4	Watchdog Circuit	
7.3 Zone	e Characteristics	
7.3.1	Zone Type	
7.3.2	Cross Alarm	
	Example	
7.3.3	Pre-Alarm	
7.3.4	Smoke Verification	
7.3.5	Zone Response Time	

Section 8 Programming ______8-1

8.1 EEPROM Information	
8.2 Downloading	
8.3 How to Use Step Programming	
8.3.1 Entering Step Programming (Mode 27)	
8.3.2 Programming Options	
8.3.3 Advancing to the Next Option	
8.3.4 Going to a Specific Step	
8.3.5 Viewing Previously Programmed Data	
8.3.6 Correcting Errors	
8.3.7 Entering Hexadecimal Digits	
8.3.8 Programming Examples	
Example 1: Choosing a Programming Option from a Menu	
Example 2: Programming Location Description Names	
8.3.9 Exiting Step Programming	
8.4 Step Programming Options	
8.4.1 Programming Steps	

Section 9 Troubleshooting

	5	
9.1	Silencing Notification Devices	9-1
9.2	Earth Ground Fault Troubleshooting	9-1

9.2.1 P3 and P4 Fault	
9.2.2 Accu-Zone, Troubleshooting (Mode 25)	
Special Notes	
9.3 Troubleshooting and System Messages	

10.1 Power Loss Reporting	
10.2 Reporting Formats	
10.2.1 SIA Format Printed Messages	
10.2.2 Silent Knight 3/1 and Sescoa 3/1 Formats	
10.2.3 Silent Knight FSK and 4+2 Formats	
10.2.4 Radionics BFSK Format	

Appendix A Programming Quick Reference

Section 1 Introduction

The Model 5204 is a low-cost fire alarm control panel with optional communicator that meets UL 864 and NFPA 72 requirements. It is available with a 12 or 24 VDC power supply, which you can select in the field. The 5204 cabinet can be surface mounted or flush mounted.

1.1 How to Use This Manual

The Model 5204 Fire Control/Communicator Installation Manual (P/N 150644) is intended for those people involved with the installation, maintenance, and programming of the 5204 panel. It covers wiring, connection to compatible products, normal operation, programming, troubleshooting, and central station reporting.

This manual is a comprehensive guide. It provides detailed instructions and can be used for reference. The installation manual is organized chronologically by the tasks that need to be performed to get the panel operating according to your needs. You can skip sections that do not apply to your installation.

In this manual, the following conventions are used:



1.2 Optional Accessories

The following Silent Knight components can be used with the Model 5204 panel.

Model	What it Does
2608 Ground Start Relay	Used for ground start phone lines(not UL listed).
4180 Status Display Module	For remote annunciation of alarm and trouble status information for each zone.
5220 Direct Connect Module	For direct alarming and trouble transmission from the 5204 to a supervising station.
5230 Remote Annunciator	Provides complete system control. Includes touchpad (keypad) with membrane keyswitches, back-lit LCD indication of zone and system status, and built-in speaker for audible annunciation. Used for programming with English-language prompts.
Quick connect program cable, part number 130294	For temporarily connecting the 5230 to the 5204 for programming.
5293 Distributed Power Module	For connecting more notification devices than the 5204 normally allows.
5541 Downloading Software	For remote programming of the 5204.
5530 Modem	Modem for downloading; required if using the 5541 software.
5205 Dialer Module	Enables the 5204 to function as a communicator panel.
7181 Zone Converter	Converts a zone from class B (style B) to class A (style D) or from class A to class B. One 7181 per zone to be converted.

Table A-1: Compatible Components (Manufactured by Silent Knight)

Circuit	12-Volt Panel	24-Volt Panel		
Primary AC	120 Vrms at 60 Hz, 2500 mA rms	120 Vrms at 60 Hz, 2500 mA rms		
Total External DC Load	3.0A	3.0A		
Accessory Power	9.5 V to 13.8 V max., 1500 mA	19.7 V to 27.6 V max., 1500 mA		
+12 V Accessory Power	8.0 V to 14.0 V, 175 mA	11.5 V to 14.0 V, 175 mA		
Bell Power	9.3 V to 13.8 V max., 1500 mA	19.8 V to 27.6 V max., 1500 mA		
Smoke Power	9.3 V to 13.8V max., 1000 mA	19.7 V. to 27.6 V. max., 1000 mA		
Battery Charging Voltage	13.5 to 13.8 V	27.0 V - 27.6 V		
Minimum Low Battery Detect	10.2 V	20.4 V		
Minimum Low AC Detect	100 Vrms at 60 Hz, full load	100 Vrms at 60 Hz, full load		
Minimum Class B Trouble Detect	1.5 mA	2.4 mA		
Maximum Class B Alarm Detect	11.7 mA	12.1 mA		
Maximum Watchdog Response Time	50 sec.	50 sec.		
<i>Note:</i> When running at full load, it is normal for the main heatsink to be hot.				

2.2 Environmental Specifications

It is important to protect the 5204 control panel from water. To prevent water damage, the following conditions should be AVOIDED when mounting the units:

- Do not mount directly on exterior walls, especially masonry walls (condensation)
- Do not mount directly on exterior walls below grade (condensation)
- Protect from plumbing leaks
- Protect from splash caused by sprinkler system inspection ports
- Do not mount in areas with humidity-generating equipment (such as dryers, production machinery)

When selecting a location to mount the 5204 control panel, the unit should be mounted where it will NOT be exposed to temperatures outside the range of 0° C-49° C (32° F-120° F) or humidity outside the range of 10%-85% at 30° C (86° F) noncondensing.

See also the mounting recommendations in Section 5.5.

2.3 Wiring Specifications

To avoid induced noise (transfer of electrical energy from one wire to another), keep input wiring isolated from high current output and power wiring. Induced noise can interfere with telephone communication or even cause false alarms. Avoid pulling one multiconductor cable for the entire panel. Instead, separate the wiring as follows:

High current input/output:	AC power, speaker, and notification device wiring
Low current input/output:	Annunciator and zone loop wiring
Audio input/output:	Telephone wiring

DO NOT pull wires from different groups through the same conduit. If you must run them together, do so for as short a distance as possible or use shielded cable. Connect the shield to circuit ground at the panel. You must route high and low voltages separately.

For the same reasons, you should route the wiring within the cabinet around the perimeter of the cabinet. It should not cross the printed circuit board where it could induce noise into the sensitive microelectronics or pick up unwanted RF noise from the high speed circuits.

High frequency noise, such as that produced by the inductive reactance of a speaker or bell, can also be reduced by running the wire through ferrite shield beads or by wrapping it around a ferrite toroid. See Figure 2-1.



Figure 2-1 Wiring Identification

Section 3 Agency Listings, Approvals, and Requirements

3.1 Federal Communications Commission (FCC)

1. If requested by the telephone company, the following information must be provided before the 5204 can be connected to the phone lines:

А	Manufacturer:	Silent Knight Security Systems
В	Model Number:	5204
С	FCC registration number	AC6USA-73710-AL-E
	Ringer equivalence:	0.9B
D	Type of jack (to be installed by the telephone company	RJ31X

- 2. This device may not be directly connected to coin telephone or party line services.
- 3. This device cannot be adjusted or repaired in the field. In case of trouble with the device, notify the installing company or return to:

Silent Knight Security Systems 7550 Meridian Circle Maple Grove, MN 55369-4927 612-493-6455 800-328-0103

- 4. If the 5204 or 5205 dialer causes harm to the telephone network, the telephone company will notify the user in advance that temporary discontinuance of service may be required. If advance notice is not practical, the telephone company will notify the user as soon as possible. Users have the right to file complaints, if necessary, with the Federal Communications Commission.
- 5. The telephone company may make changes in its facilities, equipment, operations, or procedures that could affect the operation of the equipment. If this happens, the telephone company will provide advance notice to allow you to make the necessary modifications to maintain uninterrupted service.

3.1.1 FCC Warning

Warning

This device has been verified to comply with FCC Rules Part 15. Operation is subject to the following conditions: (1) This device may not cause radio interference, and (2) This device must accept any interference received, including interference that may cause undesired operation.

3.2 Underwriters Laboratories (UL)

The 5204 is UL listed as a control unit for use in NFPA 72 systems. If the 5204 and its accessories are to be used as part of a UL installation, carefully read the UL requirements in this section. For more information on the following NFPA 72 standards, refer to the NFPA National Fire Alarm Code, 1993 Edition.

• Chapter 3

Local Protected Fire Alarm Systems

• Chapter 4

Central Station Fire Alarm Systems

Auxiliary Protected Fire Alarm Systems for Fire Alarm Service (City Box)

Remote Station Protected Fire Alarm Systems (Polarity Reversal)

3.2.1 Requirements for All Installations

General requirements are described below. When installing an individual device, refer to the specific section of the manual for additional requirements. See also the subsection below that describes special requirements for the type of installation (for example, Central Station Fire Alarm systems, Local Protected Fire Alarm systems, and so on).

- 1. All AC wiring to and from the 5204 cabinet must be enclosed in conduit.
- Total 24-hour standby current must not exceed 875 mA in 12V mode or 438 mA in 24V mode. Total 60-hour standby current must not exceed 350 mA in 12V mode or 175 mA in 24V mode.
- 3. All electrical connections must comply with the ratings shown in Section 5.6.
- 4. Because the 5204 panel itself is the main source of alarm and trouble annunciation, you must select a location for the panel that allows alarms and troubles, including pre-alarms, to be heard by end-users responsible for maintaining the panel.

3.2.2 Requirements for Central Station Fire Alarm Systems

- 1. You must program a phone number and a test time (See Section 8, Step 69 and Step 76) so that the 5204 sends an automatic daily test to the central station.
- 2. In systems using class A (style D) zones (converted using the Model 7181 Zone Converter), do not use more than 5 waterflow devices. (See Section 6.2.5.)
- 3. Auxiliary relays may NOT be programmed to activate for Pre-Alarm. (See Section 8, Step 20.)

3.2.3 Requirements for Local Protected Fire Alarm Systems

At least one UL listed supervised audible appliance must be used.

3.2.4 Requirements for Auxiliary Protected Fire Alarm Systems for Fire Alarm Service

- 1. Do not exceed the current load restrictions shown in Section 4.
- 2. The Model 5220 Direct Connect module must be installed (see Section 6.2.2 for wiring).

3.2.5 Requirements for Remote Station Protected Fire Alarm Systems - Polarity Reversal

- 1. Do not exceed the current load restrictions shown in Section 4.
- 2. The Model 5220 Direct Connect module must be installed (see Section 6.2.2 for wiring).

3.3 California Fire Marshal (CFM)

The CFM approval number for the 5204 is 7165-0559:117

3.4 Factory Mutual (FM)

The 5204 is FM approved under project # OW6A3.AY when used in conjunction with the Silent Knight Model 9000 Receiver.

3.5 Materials & Equipment Board of Acceptance (MEA)

The 5204 is now approved under MEA. Previously, approval was given from the City of New York Board of Standards and Appeals (BSA). The 5204 is now approved under MEA Number 429-92-E.

4.1 Model 5204 Wiring Diagram

Figure 4-1 is a wiring diagram for wiring the various components of the Model 5204 panel. Any device connected to terminal 24 must be UL listed for fire use, and must be rated at 12 V/24 V. Terminals 22 and 26 are the only terminals that should be used to return smoke power and should not be used for any other purpose.



Figure 4-1 Model 5204 Wiring Diagram

4.2 Current Draw Worksheet

Device	Number of Devices	Current per	Device	Standby Current	Alarm Current
For each device, use this form	la:This column	X This column	= C	urrent per number o	f devices
5204 Fire Control/	1	Standby:	120 mA	mA	
Communicator	1	Alarm:	400 mA		mA
1180 Status Display module	(2 max)	Standby:	20 mA	mA	
4100 Status Display module	(2 max.)	Alarm:	140 mA		mA
5205 Dialar	1	Standby :	10 mA	mA	
5205 Diater	1	Alarm:	100 mA		mA
5220 Direct Connect module	1	Standby:	50 mA	mA	
5220 Direct Connect module	1	Alarm:	50 mA		mA
5230 Pamota Annuncistor	(3 max)	Standby:	60 mA	mA	
5250 Remote Annunciator	(3 max.)	Alarm:	120 mA		mA
7181 Zone Converter	(1 max)	Standby 12V/24V:	52/35 mA	mA	
/181 Zolle Collverter	(4 max.)	Alarm 12V/24V:	90/65 mA		mA
		Curre	nt Subtotals:	mA	mA
Smoke Detectors	Refer to device 1	manual for current rati	ngs. See Table	es 6-2 and 6-3 for m	ax. # per loop.
		Standby:	mA	mA	
		Alarm:	mA		mA
		Standby:	mA	mA	
		Alarm:	mA		mA
		Standby:	mA	mA	
		Alarm:	mA		mA
		Curre	nt Subtotals:	mA	mA
Notification Devices	Refer to device 1	nanual for number of	devices and cur	rrent ratings.	
		Alarm:	mA		mA
		Alarm:	mA		mA
		Curre	nt Subtotals:	mA	mA
Additional Devices					
		Standby:	mA	mA	
		Alarm:	mA		mA
		Standby:	mA	mA	
		Alarm:	mA		mA
		Standby:	mA	mA	
		Alarm:	mA		mA
		Standby:	mA	mA	
		Alarm:	mA		mA
	I	Curre	nt Subtotals:	mA	mA
Total current ratings of all devi					
	ces in system (add	d A through D)*:	Γ	mA	mA

* This information must be used with Table 4-1 and Table 4-2 to complete battery calculations.

D

Е

А

В

С

4.2.1 Worksheet Example

A worksheet is included to help you calculate the amount of current the system draws on standby (idle) and in active (trouble or alarm) conditions. Refer to Table 4-2 to see the different battery sizes available and the maximum standby current load each can support.

Figure 4-2 illustrates how to complete the worksheet:

	Gross cut erties <u>Note: Do not ertie in studied areas.</u>						
العربي	row of any devices not used.			e altria di Masart	- or and a second of a second		
ſ	(b) FOR ARYONG WILL ANTER				s o carology artic plant deuteting fing an eine	ni ourrenses, entilese	
	List the number of devices	being used.	1987 6841	TE DO DO DAREES	In the prover have a process of	aradi	
	(3) The maximum number is ele-	own in parenthise	1815.				
	The number "I" printed in D	ns column indica	sei aù	- ili in missing s	urrents natings for s	no soviceo upeol.	
	that only one device can be	USAS.		1000 CBND 5000	s asvides have arts less constitues.	renti rati ngo tor	
		Number of	Curren	nt ner 🗡 🗌	Standby	Alarm	1
	Device	Devices	Dev	ice	Current	Current	
	East work devices and dischargeder	This collarse	V This	olone -	Cornerst new second	kan a Kalinaja an	۱.,
	or manuarea, as moynease.	T DES LOCATION		anana -	C. In case of party states	ar generate	Ľ
	3204 Fire Control	1	Standby:	120 mA	120 mA		
	Communicator		Alarm	400mA		400*^	
	1180 Status Display module		Standberg	20 mA			-
Ы		(2 max.)	Alarra	140 mA		mA.	1
	5205 Dialer	1	Standby:	10 mA	10 mA		
		-	Alorre	100 mA		1/00m1	
	\$220 Direct Connect module	1	Komen Barra	10 m 1	50	100-	
	3220 Direct Correct insense		secondary:	20 m/s	50 mA		
	1110 B		Aurtic	20.0hA		<u>50m</u>	
	5150 Remote Annunciator	3	Standby:	60 mA_	180 ^{ma}		
		(3 max.)	Alartic	120 mA		360 mA	
	718LZone Converier		Standby 12V/2	Am 2028 - 274	mA		
		(d march)	Alara 12V/24	N: 90.85 mA		mA.	
а.			Carren	t Subtotals:	260 mi	910m4	
1.0	Parallel Fishering	No. Constant American		i manine an article	000	CICIEI	
	smoke Detectors	Equipation and the second	other for current	r retorgo. See Joe	ter t-2 and t-2 per tea	r a broa roch	
	Model XYZ	10	Standby:	.05 mA	,1 mA		
	>	6	Alarm	1G nA		32mA	
	EGL AAOCT		Standby:	OTnA	.07mA		
	EDL 449UI		Alarme	15 nA		15mA	1
			Standby:	rs A	mA		1
			Alorro	nA		mA	1
в			Carren	r Salatatalar	$17m\lambda$	A7mt	
			Carren	Concession.	17004	477.003	
	Notification Devices	(4 max.)	Refer to dente	e manual for core	ed subage.		
		_	Alorer	nA		nA	
	\sim						ſ.,
					~		
	Total current ratings of all devices in system (add A-D)*:			360.T/hA	957 nA	1	
E.	Total garrent ratings converted to	anneres (x. 00	Co.		03604	0.957 A	1
\leq	The second second second by a second s	for the second					
	(3), in the blank spaces, write	e in any devices i	10E	_ For	row E, add the sub-	lotalo from	
	Printed on the worksheet	(smoke sleteoto	176,	(6) ra	ve A-D and multicl	y by 2001.	
	notification devices, etc.).						

Figure 4-2 Current Draw Worksheet Example

Maximum current draw for signaling devices - 1.5 A

(See Section 6.3 for additional information on signaling outputs.)

Maximum Loop resistance for smoke detectors - 30 ohms

To measure maximum loop resistance, connect an ohmmeter across the leads of a disconnected loop.

(See Table 6-2 and Table 6-3 for maximum number of smoke detectors per loop.)

4.2.2 Worksheet Requirements

The following steps must be taken when determining 5204 current ratings:

- 1. For the Model 5204, you must measure the alarm (active) current. If only one current rating is listed, the draw for that device is the same whether the system is in alarm or standby condition. The exception is for notification devices, which are rated at alarm current only. Standby current for sounding devices is 0 mA.
- 2. To measure the maximum alarm current of the panel, measure the current draw (with no devices connected to the panel) by connecting a DC amp meter in series with one of the batteries. Disconnect the AC power source. Put the panel in alarm. The meter will indicate the alarm current, which will be in the range of 120-400 mA. Fill in the system alarm current in the Current per Device column on the Current Draw worksheet. You can estimate without measuring the alarm current by filling in the maximum total alarm current of 400 mA.

Note: In a 12-volt system, measure the current from both batteries (disconnect both grounds).

- 3. For smoke detectors, notification devices and devices not mentioned in the manual, refer to the device manual for the current ratings. The worksheet example shown on the previous page provides rough estimates for a "worst case" installation.
- 4. Use Table 4-1 to determine the battery amp hour rating needed for your installation. Refer to the example (Figure 4-3) that follows. Note that the calculated rating in Row H cannot exceed the ratings shown in Table 4-2).

		Total Standby Current	Total Alarm Current
А	Total supervisory current from the Current Draw worksheet (row E).	А	
В	Number of standby hours (24 and 60 for NFPA 72, Chapter 1, 1-5.2.5).	Н	
С	Multiply Lines A and B.	AH	
D	Total alarm current from the Current Draw worksheet (row E).		А
Е	Alarm sounding period in hours. (For example, 5 minutes = .084 hours.)		Н
F	Multiply lines D and E.		AH
G	Add lines C and F.	AH	
Н	Multiply line G by 1.2. (Total ampere/hours required*)	АН	

Table 4-1: Battery Calculations

* Use next size battery with capacity greater than required.

	This calculation is based on the	Current Draw worldsheet example	ataittai.
	From this table, the inst	taller would use a 17 AH battery	
		Total Standby Current	Iotal Alarm Current
A	Total supervisory current from the Current Draw worksheet (row E).	0.360 🗚	
в	Number of standby hours (24 and 60 for NFPA 72, Chapter 1, 1-5.2.5.).	24 н	
\mathbf{C}	Multiply lines A and B.	8.64 ah	
D	Total alarm current from the Current Draw worksheet (row E).		0.957 _A
Е	Alarm sounding period in hours. (For example, 5 minutes084 hours.)		.084 н
F	Multiply lines D and E.		<i>0.08</i> .н
G	Add lines C and F.	8.72 _{AH}	
н	Multiply line G by 1.2. (Total ampere/hours required*)	10.46 ah	

Figure 4-3	Battery	Calculation	Example
J			

Warning!

Silent Knight does not support the use of batteries smaller than those listed in Table 4-2. If you use a battery too small for your installation, the system can overload it and you may have less than the required 24 hours standby power. Use Table 4-1 to calculate the correct battery amperes/hour rating needed for your installation.

5. Refer to Table 4-2 to verify the battery size you need to provide at least the total standby current you have calculated. If the installation must meet requirements for NFPA 72 (Auxiliary Protected Fire Alarm Systems for Fire Alarm Service or Remote Station Protected Fire Alarm Systems - Polarity Reversal), the total standby current cannot exceed the amount shown in the last column of the following table:

Rechargeable Battery Size	Max. Load for 24 hrs. Standby, 5 mins. Alarm	*Max. Load for 60 hrs. Standby, 5 mins. Alarm
17 Amp Hours	438 mA	175 mA
34 AH (if wired in parallel)	875 mA	350 mA

Fable 4-2: Maximum	Battery	/ Standby	/ Load
--------------------	---------	-----------	--------

* Required for NFPA 72 Auxiliary Protected Fire Alarm systems for Fire Alarm Service (City Box) and Remote Station Protected Fire Alarm systems (Polarity Reversal).

The following formula was used to calculate the figures in Table 4-2:

I =	[AH÷	H] x ().70
Where:	Ι	=	Standby current
	AH	=	Ampere-hour rating of battery
	Н	=	Standby hours
	0.70	=	A constant used to de-rate the battery to assure a 5-year life.

6. Ensure that the total alarm current you calculated, including current for the panel itself, does not exceed 3.5 A. This is the maximum alarm current allowable, whether the panel provides 12 V or 24 V of smoke power.

4-6

Section 5 Control Panel Installation

The major components of the Model 5204 PC board are described in this section. Figure 5-1 shows the 5204 (fuseless) printed circuit board.



Figure 5-1 Model 5204 Board Assembly

Caution

To avoid the risk of electrical shock, Do NOT apply power to the Model 5204 until told to do so in this manual (See Note in Section 6.2).

5.1 Grounding the Model 5204 Cover

Before connecting power to the 5204, connect the earth ground wire to the base and cover. Make sure that the ring lugs are oriented properly. The proper connection and orientation are shown in Figure 5-2. The star washers must be located between the ring lugs and the painted surfaces.

After attaching the cover and base, make a slight bend in the wire attached to the cover. This is to keep it from being caught between the cover and base when the cover is closed.



Figure 5-2 Connecting the Ground Wire

5.2 Smoke Power Selection

With AC power removed and batteries removed, insert the supplied jumper block (P/N 130412) into P5 for 24V or P6 for 12V.

5.3 Power Supply Wiring

A transformer is used to supply 31 VAC (220 VA) to power the system under normal conditions and to supply charging current to the backup batteries. The primary winding must be connected directly to the 120 VAC, 60 Hz power source (unswitched). Connect the secondary to the 5204 by plugging the cable into the AC connector on the circuit board.

Note: It may be necessary for a professional electrician to connect the pigtail wires on the primary winding to the 120 VAC source.

5.4 Battery Connection

Note: When using two batteries, it is recommended that they be of the same ampere hour (AH) rating and approximately the same age.

Battery cable connectors enable installation of one or two 12 VDC, 17 A rechargeable batteries. Two sets of battery leads are provided for battery connection. When connecting a single battery, connect one of the red leads to the positive side of the battery. Connect a black lead to the negative side of the battery.

If using a second battery, connect the remaining lead to the positive side of the second battery. Connect the remaining black lead to the negative side of the second battery.

5.5 Mounting the 5204

Read the environmental specifications in Section 2.2 before mounting the 5204 panel.

The panel should be accessible to "Main Drop" wiring runs. The 5204 panel should be mounted as close to the center of the building as possible and located within a secured area, but should be accessible for testing and service. End-users responsible for maintaining the panel should be able to hear alarms and troubles. When selecting a location, keep in mind that the panel itself is the main source of alarm and trouble annunciation.

Mount the 5204 so it is firmly secured to the wall surface. When mounting the 5204 on concrete, especially when moisture is expected, attach a piece of 3/4-inch plywood to the concrete surface and then attach the 5204 to the plywood. Also mount any other desired components (such as external printer) to the plywood. If you will be flush mounting the cabinet, the hole for the enclosure should be $14 \ 1/2$ " x 19 1/8" (width x length of box only). Do NOT flush-mount in a wall designated as a fire break.

5.6 Terminal Strip Description

The terminal strips on the PC board are nonremovable. Table 5-1 below lists the function and electrical rating of each terminal. Note the following:

- The total load of all devices attached to the system must not exceed 3.0 A.
- Alarm polarity is shown for bells (terminals 7-10). Normal polarity is the opposite.
- Terminals 22 and 26 are the only terminals that should be used to return smoke power, and they should not be used for any other purpose.

Important!

The 5204 emits a hum that is not noticeable to most end users unless they are near the panel in a very quiet environment.

Terminal Number	Terminal Description	Nominal VDC Output ("System Normal" Condition)	
Humber		12 V Mode	24 V Mode
1*	Auxiliary Power (+) - 1500 mA max.	13.65	27.3
2*	Ground	0	0
3*	Annunciator Power (+) - 500 mA max.	13.5	13.6
4*	Serial Annunciator Data Out (SKO)	9.1	9.2
5*	Serial Annunciator Data In (SKI)	6.6	6.7
6*	External Silence Switch or Alarm Reset	8.6	8.6
7	Bell 1 - 1500 mA max.	5.0	10.0
8	Bell 1 +	0.95	1.9
9	Bell 2 - 1500 mA max.	5.0	10.0
10	Bell 2 +	0.95	1.9
11	Relay 1 Normally Open	N/A	N/A
12	Relay 1 Common	N/A	N/A
13	Relay 1 Normally Closed	N/A	N/A
14	Relay 2 Normally Open	N/A	N/A
15	Relay 2 Common	N/A	N/A
16	Relay 2 Normally Closed	N/A	N/A
17*	Auxiliary Power - 175 mA	13.64	13.77
18*	Zone 1 (Class B/Style B) Input	0.08	0.16
19*	Smoke Power	13.65	27.3
20*	Zone 2 (Class B/Style B) Input	0.08	0.16
21*	Smoke Power	13.65	27.3
22*	Ground	0	0
23*	Zone 3 (Class B/Style B) Input	0.08	0.16
24*	Smoke Power	13.65	27.3
25*	Zone 4 (Class B/Style B) Input	0.08	0.16
26*	Ground	0	0

Table 5-1: Terminal Strip Description

* Power-limited

Note: Combined smoke power maximum capacity is 1000 mA.

5.7 Model 5205 Dialer and Telephone Line Connection (Optional)

The Model 5205 Dialer Module enables the 5204 to function as a communicator panel and provides the following features:

- Optional two-number dialing with same or different account codes and reporting formats. Alarms, troubles, disables, and tests can be programmed to report to either or both numbers.
- Programmable as rotary-only or as rotary/Touch-Tone dialing.
- Ring Detect feature on line 1 for downloading data to panel from a remote computer site.
- Transient voltage protection of phone lines.
- Automatic daily test (programmable from Model 5230 annunciator, built-in touchpad, or remote site via downloading option).
- Optional ground start operation (not for use on UL systems).
- Compatibility with the following UL fire listed receivers:

Receiver	Formats it will Receive	
Silent Knight Model 9000	All formats listed in Section 10	
Osborn & Hoffman Quickalert	All formats listed in Section 10	
Ademco 685	All tone burst formats (3/1 1400 Hz)	
FBI CP220	3/1 and 4 + 2 formats	
Radionics D6500	BFSK 1400/2300 formats	

Installation

To meet NFPA 72 Central Station Fire Alarm Systems requirements, both telephone lines must be installed.

Connect the 5205 to the phone line using an RJ31X type phone jack. The telephone company will install an RJ31X jack upon request.

The 5205 comes with stand-offs that you can place into the four holes just left of the built-in touchpad on the 5204 panel. To connect the 5205 to the 5204, make sure the dialer connector



pins are positioned correctly before pressing the 5205 onto the stand-offs.



Ring Detect Circuit

If the installing company calls the 5204 to up- or download data to or from a remote computer, the built-in ring detect circuit on line 2 will detect the ring. After the programmed number of rings (Step 55 in Section 8), it seizes the line and allows the transfer of data.

The 5204 has built-in dual phone line monitors. These circuits will detect any fault in the phone lines by monitoring the DC voltage present on the lines. They feature a delay of approximately 40-90 seconds before a line fault is reported as a trouble. When a fault is detected, the audible trouble signal will sound and the trouble will be reported to the central station over the remaining phone line.

A situation could occur where both phone lines appear to be good, but the dialer cannot get through to the central station on the first line. In this case, the 5204 will switch phone lines and attempt the call again using the second line.

Note: To comply with industry standards, this product is equipped with line seizure. Any time the system's dialer needs to communicate with the central station, it will not be possible to use any telephones that are on the same line(s) as the fire system. Normally, this condition will last approximately one minute, but under adverse telephone circuit conditions, could last for as long as 15 minutes.

5.8 Cable Connectors

Status (P1)

Connects the Model 4180 display model to the 5204.

Model 5230 (P2)

Can be used to *temporarily* connect the Model 5230 Remote Annunciator to the 5204 for programming or troubleshooting.

Note: A quick connect program cable (P/N 130294) can be ordered separately for this connection.

Warning!

Do NOT use connector P2 for permanent installation. If the annunciator is to be installed permanently, it MUST be wired to the 5204 terminal block (see Section Wiring the 5230 Remote Annunciator).

Power Supply (AC) Connector (P4)

Connects the 5204 control panel to the power supply.

6.1 Zone Wiring

This manual refers to fire zone types using the latest NFPA standard designations. If you have questions about the class or style, refer to the *NFPA 72 National Fire Alarm Code*, 1993 *Edition*.

Note: For purposes of this manual, a normally open device is one with contacts that conduct when in the alarm condition and do not conduct in the non-alarm condition.

The 5204 features four fully supervised, class B (style B) fire zones (also known as loops). All four zones have ground-fault detection and are protected against transient voltages.

Each zone consists of a two-wire circuit that detects the occurrence of an open in the loop, but may not be able to detect an alarm after such an occurrence. A short across the EOL resistor of the loop will cause an alarm to sound and the 5204 will report the trouble to the central station (if programmed to do so). An open or short to ground is a trouble condition. Use only normally open initiating devices for class B (style B) fire zones.

Zones 17 through 26 are class B (style B) fire zones. Figure 6-1 shows how to wire a class B loop. One side of each class B loop will connect to a zone input terminal and the other side of each loop will connect to smoke power. At the end of each class B (style B) loop, you must install a Model 7628 4.7K-ohm EOL resistor.





6.1.1 Four-Wire Smoke Detector Connection

Figure 6-2 illustrates how a UL listed four-wire smoke detector must be connected to a class B (style B) zone.

When wiring a four-wire smoke detector to class B (style B) zones, you must use a power supervision unit. The recommended device is an ESL 204 -12/24 V. The 7628 EOL resistor and the ESL 204 must be installed at the last detector in the loop.



To determine the maximum loop wiring resistance, connect an ohmmeter across the leads of a disconnected loop.

Figure 6-2 Four-Wire Smoke Detector Wiring

Table 6-1 shows other four-wire smoke detectors that can be used with the 5204 panel:

Manufacturer	Model Name/Number	12 or 24 Volt Panel	
Detection Systems	DS200/DS200HD	Both	
Detection Systems	MB200	Both	
ESI	445 Series	Both	
LSL	449 Series	Both	
	624	24	
	812	12	
GENTEX	824	24	
	2040-12 Power Supervision Unit	12	
	2040-24 Power Supervision Unit24	24	
	1851B	Both	
System Sensor	2851/2851BTH	Both	
	DH400ACDC	24 or AC	

Table 6-1: Compatible Four-Wire Smoke Detectors

6.1.2 **Two-Wire Smoke Detector Connection**

Figure 6-3 shows how to connect two-wire smoke detectors to class B (style B) zones.



Figure 6-3 Two-Wire Smoke Detector Wiring

Table 6-1 and show the two-wire smoke detectors that are approved for use with the 5204.

Notes for Both Tables

- 1. If a separate base is used with a detector, the model number is shown in parentheses in the Model column.
- 2. In the Type column, I = Ionization, P = Photoelectric, D = Duct
- 3. ID = Detector (Base) Identifiers
- 4. Control unit Smoke Reset Time must be programmed for a number greater than or equal to the maximum reset time of the smoke detector (last column of chart).
- 5. The maximum number of smoke detectors per zone is determined by both the current draw and the impedance of the smoke detector. If too many smoke detectors are used on any zone, false alarms could occur.
- 6. If different models of detectors are mixed on a zone, false alarms could occur.

Note: The 5204 contains a programmable smoke reset time. Be sure to program the panel to meet the reset time of the detectors.

Manuf.	Model	Туре	*ID	5204 (Max. per Loop)	Smoke Det.Reset Time
	DS200 (MB200-2W)	Р	D	15	1 sec.
Detection Systems	DS200HD (MB200-2W)	Р	D	15	1 sec.
	DS250 (MB2W or MB2WL)	Р	B (A)	11	1 sec.
	DS250TH (MB2W or MB2WL)	Р	B (A)	11	1 sec.
	DS250HD (MB2W or MB2WL)	Р	B (A)	11	1 sec.
	425C	Р	S10	20	1 sec.
	425CT	Р	S10	20	1 sec.
	425CR	Р	S10	20	1 sec.
	425CRT	Р	S10	20	1 sec.
	429C (S10A)	Р	S10A	12	1 sec.
	429CT (S10A)	Р	S10A	12	1 sec.
	429CRT (S11A)	Р	S11A	12	1 sec.
	429CST (S11A)	Р	S11A	12	1 sec.
EGI	611U (601U)	Р	S10 (S00)	24	1 sec.
ESL	611UD (601U)	D	S10 (S00)	24	1 sec.
	611UT (610U)	Р	S10 (S00)	24	1 sec.
	612U (601U)	Ι	S10 (S00)	24	1 sec.
	612U (601U)	D	S10 (S00)	24	1 sec.
	612UD (601U)	Р	S10 (S03)	24	1 sec.
	611U (602U)	D	S10 (S03)	24	1 sec.
	611UD (602U)	Р	S10 (S03)	24	1 sec.
	611UT (602U)	Ι	S10 (S03)	24	1 sec.
	612UD (602U)	D	S10 (S03)	24	1 sec.
	1451 (B401B)	Ι	А	10	6 sec.
	1800	Ι	Α	12	0.9 sec.
	1851B (B101B)	ID	Α	10	2 sec.
	1851DH (DH1851DC)	ID	Α	10	2 sec.
	2151 (B110LP)	Р	А	10	.3 sec.
	2400	Р	Α	10	6 sec.
	2400 (DH400)	Р	А	10	1 sec.
System Sensor	2400TH	Р	A	10	6 sec.
	2451 (B401B)	Р	Α	10	1 sec.
	2451TH (B401B)	Р	Α	10	6 sec.
	2800	Р	Α	10	6 sec.
	2800TH	Р	А	10	6 sec.
	2851B (B101B)	PD	A	10	2 sec.
	2851TH (B101B)	PD	A	10	2 sec.
	2851DH (DH2851DC)I	PD	А	10	2 sec.

Voltage range: 9.5 VDC - 14 VDC: Identifier: 12C

Note: Note: The 5204 *contains a programmable smoke reset time. Be sure to program the panel to meet the reset time of the detectors.*

Manuf.	Model	Туре	*ID	5204 (Max. per Loop)	Smoke Det. Rest Time
A	55000-250 (45681-200)	Ι	55000-250 (45681-200)	24	1 sec.
Аропо	55000-350 (45681-200)	Р	55000-350 (45681-200)	12	1 sec.
	1451 (B401B)	Ι	А	12	6 sec.
	1800	Ι	А	12	0.9 sec.
	1851B (B101B)	ID	А	12	2 sec.
	1851DH (DH1851DC)	ID	А	12	2 sec.
	2151 (B110LP)	Р	А	12	.3 sec.
	2400	Р	А	12	6 sec.
	2400 (DH400)	Р	А	12	1 sec.
System Sensor	2400TH	Р	А	12	6 sec.
	2451 (B401B)	Р	А	12	1 sec.
	2451TH (B401B)	Р	А	12	6 sec.
	2800	Р	А	12	6 sec.
	2800TH	Р	А	12	6 sec.
	2851B (B101B)	PD	А	12	2 sec.
	2851TH (B101B)	PD	А	12	2 sec.
	2851DH (DH2851DC)I	PD	А	12	2 sec.
Detection	DS200 (MB200-2W)	Р	D	24	1 sec.
Systems	DS200HD (MB200-2W)	Р	D	24	1 sec.
-	425	Р	S10	30	1 sec.
	425CT	Р	S10	30	1 sec.
	429C (S10A)	Р	S10A	14	1 sec.
	429CT (S10A)	Р	S10A	14	1 sec.
	429CRT (S11A)	Р	S11A	14	1 sec.
	429CST (S11A)	Р	S11A	14	1 sec.
	611U (601U)	Р	S10 (S00)	30	1 sec.
ESL	611UD (601U)	D	S10 (S00)	30	1 sec.
	611UT (610U)	Р	S10 (S00)	30	1 sec.
	612U (601U)	I	S10 (S00)	30	1 sec.
	612U (601U)	D	S10 (S00)	30	1 sec.
	612UD (601U)	D	S10 (S03)	30	1 sec.
	611UD (602U)	Р	S10 (S03)	30	1 sec.
	611UT (602U)	I	S10 (S03)	30	1 sec.
	612UD (602U)	D	S10 (S03)	30	1 sec.
Gentex	224	Р	-25-1	16	6 sec.
TT 1.1.	SLK-24F (HS-224D)	Р	HD-3 (HB-5)	20	0.1 sec.
HOCHIKI	SLK-24FH (HS-224D)	Р	HD-3 (HB-5)	20	0.1 sec.

Voltage range: 9.5 VDC - 14 VDC: Identifier: 12C

6.2 Connections to Compatible Silent Knight Products

This section describes the connections of the following Silent Knight products:

- Model 4180 Status Display Module (see Section 6.2.1)
- Model 5220 Direct Connect Module (see Section 6.2.2)
- Model 5230 Remote Annunciator (see Section 6.2.3)
- Model 5395 Distributed Power Module (see Section 6.2.4)
- Model 5205 Dialer Module (see Section 5.7)
- Model 7181 Zone Converter (see Section 6.2.5)
- Note: Once you have installed the 5204 and, if applicable, the 5230 and the 4180, test the basic system. Apply power, test the touchpad, then remove the power. Wire each auxiliary device with the power off. After you install each device, test it by re-applying the power. When you power up the 5204, the two dots on the built-in touchpad display will alternately flash on and off.

Note also that there is a 2-second power-up delay on the 5230.

6.2.1 Model 4180 Status Display Module

The Model 4180 Status Display module provides remote annunciation of alarm and trouble status information for each zone.

The 4180 has 2 connectors, each of which has 8 outputs available for annunciation. These outputs are active high at +12 VDC. Each output can provide up to 100 mA of current, with a total limitation of 175 mA (when used with the 5204). The module has 4 normally open relays that are nondedicated, and therefore can be wired to be active with any of the outputs. The 4180 is not supervised. Table 6-4 shows the system status indicated by each LED.

Table 6-4: Model 4180 Connection

Do not use the 4180 relays in a 12 V 5204 installation.

Connector P2	System Status
1	Alarm 1
2	Alarm 2
3	Alarm 3
4	Alarm 4
5	Trouble 1
6	Trouble 2
7	Trouble 3
8	Trouble 4

Connector P3 System Status Line #1 Trouble 1 2 Line #2 Trouble 3 Bell #1 Trouble 4 Bell #1 Trouble 5 Battery Trouble 6 AC Trouble 7 Silence Trouble 8 Dialer Trouble

The 4180 can be used to interface to long-range RF systems.



Figure 6-4 Model 4180 Connection

When using a 4180, maintain a physical separation of one-half inch or more between field wires and connection points to prevent damage from transients.

6.2.2 Model 5220 Direct Connect Module

The 5220 Direct Connect Module can be used with the 5204 to meet NFPA 72 Remote Signaling or Local Protective Signaling standards. The 5220 requires four connections to the 5204 and provides outputs for direct connect (city box) and polarity reversal.

To meet the 60-hour standby power requirements for NFPA 72 systems, normal standby currents are de-rated. See Section 4.2 for these current values.

6.2.2.1 Installation

Locate the knockout on the right side of the 5204 cabinet to connect the 5220 using a short piece of conduit (must not exceed 20 feet in length).

A four-wire pigtail is provided to wire the 5220 to the 5204. Figure 6-5 shows how to wire the Model 5220 Direct Connect module. The wiring chart uses bell #2 as the initiating loop. Program bell #2 to be active for the events to be reported.



Figure 6-5 Model 5220 Wiring Diagram

6.2.2.2 City Box Connect (24 VDC Systems Only)

(For NFPA 72 Auxiliary Protected Fire Alarm systems for fire alarm service.)

With the 5220, you can connect the 5204 directly to a municipal fire alarm box or "city box." The city (master) box is an enclosure that contains a manually operated transmitter used to send an alarm to the municipal communication center, which houses the central operating part of the fire alarm system. To ensure communication of an active alarm status, use the 5220 only with 5204 24 V systems when connected to a series type DC master box.

Wire the 5220 to the 5204 as shown in Figure 6-5. Wire the city box coil to terminals 3 and 4 in the 5220. Maximum coil and wire resistance (combined) is 30 ohms.

It is not possible to reset the remote indication until you clear the condition and reset the 5204 panel.

Select relay 2 for 5220 city box. When you select 5220 operation, bell 2 and relay 2 cannot be used for any other purpose.

Any zone programmed to activate bell 2 will cause an alarm to be sent.

6.2.2.3 NFPA 72 Polarity Reversal (12 or 24 VDC Systems)

The 5220 provides a current that reverses polarity during an alarm or removes current during a trouble condition.



Wire the 5220 for polarity reversal as shown in Figure 6-6.

Figure 6-6 Wiring the 5220 for Polarity Reversal

Alarms will override trouble conditions, and it will not be possible to reset the remote indication until you clear the condition and reset the 5204 panel.

Select relay 2 for 5220 Direct Connect. When you select 5220 operation, bell 2 and relay 2 cannot be used for any other purpose.

Any trouble condition will cause a trouble to be sent. Any zone programmed to activate bell 2 will cause an alarm to be sent.

6.2.3 Model 5230 Remote Annunciator

The Model 5230 Remote Annunciator is an optional touchpad (keystation) you can use for English-language programming. The 5230 also provides trouble and alarm information.

When programming the 5204, be sure to select the correct number of supervised annunciators (see Section 8.4.3).

6.2.3.1 Setting ID Codes

Before permanently installing the Model 5230 Remote Annunciator, you must first set its identification codes. Each annunciator to be supervised must be given its own identification codes. The ID numbers must start at 1 and progress sequentially to 3 (3 annunciators max.). Upon initial power up, the address of each annunciator is displayed.

On the back of each annunciator is a small 4-position dip switch you can use to set the ID code. Table 6-5 shows the positions (up or down) of the various switches for specific ID codes.

ID Number	Switches				
	1	2	3	4	
0 *	Up	Up	Up	Up	
1	Down	Up	Up	Up	
2	Up	Down	Up	Up	
3	Down	Down	Up	Up	

Table 6-5: Model 5230 Dip Switch Settings

* Not Supervised

6.2.3.2 Wiring the 5230 Remote Annunciator

A 4-position terminal block is provided with the Model 5230 Annunciators to connect them to the 5204. Figure 6-7 shows the wiring for the Model 5230.



Figure 6-7 Model 5230 Connection
6.2.3.3 Mounting the 5230 Remote Annunciator

For UL installations, the 5230 Remote Annunciators must be mounted on a dual gang electrical box.

To mount the annunciator, you must first remove the rear mounting plate.

To do this, insert a #4 flat blade screwdriver into the slots located on the bottom edge of the annunciator. Gently turn the screwdriver until the mounting plate pulls away from the frame. Once you remove the mounting plate, you can secure it to the wall using #6 or #8 screws. The mounting plate should be oriented so that the word TOP is toward the top of the plate and facing you. Through the square hole in the mounting plate, run the wiring to the annunciator.

When all of the wires are connected to the annunciator, set the top of the annunciator over the tabs on the top of the mounting plate. Make sure the wires are not pinched between the frame and the mounting plate. Press each corner of the bottom side onto the annunciator mounting plate until you hear it click into place.

6.2.4 Model 5395 Distibuted Power Module

Figure 6-8 shows you how to connect the Model 5395 to the Model 5204 panel.



Figure 6-8 Model 5295 Connection

Note: You may have to gently squeeze the annunciator (top to bottom) to align it while snapping the bottom edge into place.

6.2.5 Model 7181 Zone Converter

The Model 7181 Zone Converter adapts the 5204 class B (style B) zones so that they can be connected to class A (style D) initiating devices. Figure 6-9 shows a typical installation. Refer to the Model 7181 Installation Manual (P/N 150632) for further information.



Figure 6-9 Connecting Class B (Style B) Sensor to Class A (Style D) Panel (24 V)

6.3 Supervised Notification Device Outputs

Note: To reduce the possibility of false alarms and transient damage, DO NOT bundle telephone wires together with notification device or zone wires.

The 5204 provides two supervised notification device outputs to annunciate alarm conditions. These outputs can be programmed for each individual zone. For proper operation, you must use polarized notification devices with a model 7628 4.7K ohm end-of-line (EOL) resistor on each loop. See Figure 6-10 for connection to the 5204 panel.



Figure 6-10 Model 5204 Notification Device Connections

The UL listed sounding appliances that can be used with the 5204 are listed in Table 6-6 and Table 6-7.

Manufacturer	Model Number	Device Type
Faderal Signal	VALS	Strobe
rederar Signar	450-D	Horn
	HG124	Horn
Gentex	SHG12L	Horn Strobe
	SHG12H	Horn Strobe
	34T-12-R	Alarm Horn
	462-G10-12-R	Bell
	7001T-12-R	Mini-Horn
	7001T-12-W	Mini-Horn
	7001T-12W-FR	Strobe Horn
	7002T-12-W-FR	Strobe Horn
	MB-G6-12-R	Motor Bell
XX71 1 1	MB-G10-12-R	Motor Bell
	MBS-G6-12-W-HF-R	Motor Bell with Strobe
Wheelock	MBS-G10-12-W-HF-R	Motor Bell with Strobe
	MIZ-12-R	Mini-Horn
	MIZ-12-W	Mini-Horn
	MIZ-12-WS-VF-R	Mini-Horn/Strobe
	MT-12/24-R	Strobe Horn
	V7001T-W-FR	Strobe Horn
	WST-12-FR	Strobe
	WS1T-12-FR	Strobe
	WS3T-12-FR	Strobe

Manufacturer	Model Number Device Type	
	446X 12/24VDC	Vibrating Bell
	476X 12/24VDC	Vibrating Bell
	477X 12/24VDC	Single Stroke Bell
	5303B-0-14-()-DC	Chime (flush)
	5304B-0-14-()-DC	Chime (surface)
	5305B-0-4-()-DC	Chime (ceiling)
	5306B-0-14-()-24-DC	Chime/Strobe (flush)
	5307B-0-14-()-24-DC	Chime/Strobe (surface)
	5308B-0-4-()-24-DC	Chime/Strobe (ceiling)
	5333B-0-14-24-DC	Multi-Tone Horn (flush)
	5334B-0-14-24-DC	Multi-Tone Horn (surface)
	5336B-()-14-24-DC	Multi-Tone Horn/Strobe (flush)
	5337B-()-14-24-DC	Multi-Tone Horn/Strobe (surface)
	5338B-()-4-24-DC	Multi-Tone Horn/Strobe (ceiling)
	5343B-0-14-24-DC	Single Tone Horn/Strobe (flush)
	5344B-0-14-24-DC	Single Tone Horn/Strobe (surface)
Faraday	5345B-0-4-24-DC	Single Tone Horn/Strobe (ceiling)
1 uruduy	5348B-()-4-24-DC	Single Tone Horn/Strobe (ceiling)
	5373B-0-14-(12 or 24)-DC	8-Tone Horn/Strobe (flush)
	5374B-0-14-(12 or 24)-DC	8-Tone Horn/Strobe (surface)
	5375B-0-4-(12 or 24)-DC	8-Tone Horn/Strobe (ceiling)
	5376B-0-14-24-DC	8-Tone Horn/Strobe (flush)
	5377B-0-14-24-DC	8-Tone Horn/Strobe (surface)
	5378B-0-4-24-DC	8-Tone Horn/Strobe (ceiling)
	5405B-0-14-24-DC	Sync Control Unit
	5508B-()-14-24-DC	Single Gang Sync Strobe (flush)
	5521B-()-14-24-DC	4" Square Sync Strobe (surface)
	5522B-()-14-24-DC	4" Square Sync Strobe (flush)
	6126B-U-14-24 VDC	Horn/Strobe
	6223B-0-14-24-DC	Horn (flush)
	6224B-0-14-24-DC	Horn (surface)
	6225B-0-4-24-DC	Horn (ceiling)
	6226B-()-14-24-DC	Horn/Strobe (flush)
	6227B-()-14-24-DC	Horn/Strobe (surface)

Table 6-7: Compatible 24-Volt Notification Devices

Manufacturer	Model Number	Device Type
	6228B-()-4-24-DC	Horn/Strobe (ceiling)
	6243B-0-14-24-DC	Electron-Mechanical Horn (flush)
	6244B-0-14-24-DC	Electron-Mechanical Horn (surface)
	6245B-0-4-24-DC	Electron-Mechanical Horn (ceiling)
	6246B-()-14-24-DC	Electron-Mechanical Horn/Strobe (flush)
	6247B-()-14-24-DC	Electron-Mechanical Horn/Strobe (surface)
	6248B-()-4-24-DC	Electron-Mechanical Horn/Strobe (ceiling)
Faraday (continued)	6300B-0-14-24-DC	Mini-Horn (flush)
	6301B-0-14-24-DC	Mini-Horn (surface)
	6302B-()-4-24-DC	Mini-Horn (ceiling)
	6310B-0-14-24-DC	Mini-Horn/Strobe/Strobe (flush)
	6311B-0-14-24-DC	Mini-Horn/Strobe/Strobe (surface)
	6312B-()-14-24-DC	Mini-Horn/Strobe/Strobe (ceiling)
	6320B-0-14-24-DC	Sync Mini Horn/Strobe (1 gang)
	6321B-0-14-24-DC	Sync Mini Horn/Strobe (1,2 gang)
	6322B-()-14-24-DC	Mini Horn/Sync Strobe (1,2 gang, 4SQ)
Federal Signal	450	Horn
	VALS	Horn/Strobe
	GX90-4	Horn
	GXS-4-15-1	Strobe
	GXS-4-1575	Strobe
	GX90S-4-15	Horn
	GX90S-4-1575	Horn
	HG124	Horn
Gentey	SHG24-1575	Horn/Strobe
Gentex	SHG24-15	Horn/Strobe
	GMH-24-X	Horn
	GMS-24-X	Horn/Strobe
	GMS-24-X	Horn/Strobe
	G0T24	Horn
	G0S24-X	Horn
	WGMS-24-X	Horn/Strobe

Table 6-7:	Compatible	24-Volt	Notification	Devices
	companyie	24-101	Notification	DEVICES

Manufacturer	Model Number	Device Type	
	MASS241	Horn/Strobe	
	MASS24110ADA	Horn/Strobe	
	MASS2415ADA	Horn/Strobe	
	MASS2475ADA	Horn/Strobe	
	SS1215ADA	Strobe	
System Sensor	SS4110ADA	Strobe	
System Sensor	SS2415ADA	Strobe	
	SS2475ADA	Strobe	
	PS2415ADA	Mini-Horn/Strobe	
	PS241575ADA	Mini-Horn/Strobe	
	PS24110ADA	Mini-Horn/Strobe	
	PS2475ADA	Mini-Horn/Strobe	
	46T-G4-24-R	Bell	
	46T-G6-24-R	Bell	
	46T-G10-24-R	Bell	
	46T-G6-24-WS-24-HF-R	Strobe/Bell	
	46T-G10-24-WS-24-HF-R	Strobe/Bell	
	46T-G6-24-WH-24-HF-R	Strobe/Bell	
	46T-G10-24-WH-24-HF-R	Strobe/Bell	
	7001T-12\24-W-FR	Strobe Horn	
	7002T-12\24-W-FR	Strobe Horn	
	AES-DL1-R	Multitone Horn	
	AES-EL1-R	Multitone Horn	
Wheelock	AES-DL1-WS-24-VF-R	Multitone Horn	
WINCHOCK	AES-EL1-WS-24-VF-R	Multitone Horn	
	AES-DL1-WH-24-VF-R	Multitone Horn	
	AES-EL1-WH-24-VF-R	Multitone Horn	
	AES-DL1-WM-24-VF-R	Multitone Horn	
	AES-EL1-WM-24-VF-R	Multitone Horn	
	AH-24-R	Horn	
	AMT-12\24-R	Strobe Horn	
	AMT-24-LS-VFR	Strobe Horn	
	AMT-24-LSM-VFR	Strobe Horn	
	AMT-24-IS-VFR	Strobe Horn	
	AS-2415-VFR	Strobe Horn	
	AS-241575-VFR	Strobe Horn	

Table 6-7: Com	patible 24-Vol	t Notification	Devices

Manufacturer	Model Number	Device Type
	AS-2430-VFR	Strobe Horn
	AS-2475-VFR	Strobe Horn
	AS-24110-HFR	Strobe Horn
	SM-12\24-R	Strobe Horn Controller
	DSM-12\24-R	Strobe Horn Controller
	CF-BF1	Chime
	CF-BF1-R	Chime
	CH-CF1	Chime
	CH-CF1-R	Chime
	CH-CF1-W	Chime
	CH-DF1	Chime
	CH-DF1-R	Chime
	CH-BF1-WS-24-HF-R	Strobe Chime
	CH-CF1-LS-24	Strobe Chime
	CH-CF1-MS-24	Strobe Chime
	CH-CF1-IS-24	Strobe Chime
	CH-CF1-LS-24-CFW	Strobe Chime
XX71 1 1 - (CH-CF1-MS-24-CFW	Strobe Chime
wheelock (cont.)	CH-CF1-IS-24-CFW	Strobe Chime
	CH-CF1-WS-24-CF-W	Strobe Chime
	CH-DF1-LS-24	Strobe Chime
	CH-DF1-MS-24	Strobe Chime
	CH-DF1-IS-24	Strobe Chime
	CH-DF1-LS-24-VFR	Strobe Chime
	CH-DF1-LSM-24-VFR	Strobe Chime
	CH-DF1-MS-24-VFR	Strobe Chime
	CH-DF1-IS-24-VFR	Strobe Chime
	CH-DF1-WM-24-VFR	Strobe Chime
	CH-DF1-WS-24-VF-R	Strobe Chime
	DSM-12/24	Sync Module
	EH-DL1-R	Electronic Horn
	EH-EL1-R Electronic Horn	Electronic Horn
	EHS-DL1-W-VF-R	Strobe Horn (single input)
	EHS-EL1-W-VF-R	Strobe Horn (single input)
	EH-DL1-WS-24-VF-R	Strobe Horn (dual input)
	EH-EL1-WS-24-VF-R	Strobe Horn (dual input)

Table 6-7: Compat	ible 24-Volt I	Notification	Devices
-------------------	----------------	--------------	---------

Manufacturer	Model Number	Device Type
	EH-DL1-WH-24-VF-R	Strobe Horn (dual input)
	EH-EL1-WH-24-VF-R	Strobe Horn (dual input)
	EH-DL1-WM-24-VF-R	Strobe Horn (dual input)
	EH-EL1-WM-24-VF-R	Strobe Horn (dual input)
	HSW-24-HFR	Remote Strobe
	HS2W-24-HFR	Remote Strobe
	HSPW-24-HFR	Remote Strobe
	IS-24-VFR	Remote Strobe
	IS1-24-VFR	Remote Strobe
	IS3-24-VFR	Remote Strobe
	ISP-24-HFR	Remote Strobe
	LS-24-VFR	Remote Strobe
	LS1-24-VFR	Remote Strobe
	LS3-24-VFR	Remote Strobe
	LSP-24-HFR	Remote Strobe
	LSM-24-VFR	Remote Strobe
	LS1M-24-VFR	Remote Strobe
Wheelock (cont.)	LS3M-24-VFR	Remote Strobe
wheelock (cont.)	LSPM-24-VFR	Remote Strobe
	MS-24-VFR	Remote Strobe
	MS1-24-VFR	Remote Strobe
	MS3-24-VFR	Remote Strobe
	MSP-24-HFR	Remote Strobe
	MB-G6-24-R	Motor Bell
	MB-G10-24-R	Motor Bell
	MBS-G6-24-W-HF-R	Motor Bell with Strobe
	MBS-G10-24-W-HF-R	Motor Bell with Strobe
	MIZ-24-R	Mini-Horn
	MIZ-24-W	Mini-Horn
	MIZ-24-LS-VFR	Mini-Horn/Strobe
	MIZ-24-LSM-VFR	Mini-Horn/Strobe
	MIZ-24-MS-VFR	Mini-Horn/Strobe
	MIZ-24-HSW-HFR	Mini-Horn/Strobe
	MIZ-24-IS-VFR	Mini-Horn/Strobe
	MIZ-24-WS-VF-R	Mini-Horn/Strobe
	MIZ-24-WS-VF-W	Mini-Horn/Strobe

Table 6-7: Compatible 24-Volt Notification Devices

Manufacturer	Model Number	Device Type
	MIZ-24-WH-VF-W	Mini-Horn/Strobe
	MIZ-24-WM-VF-W	Mini-Horn/Strobe
	MT-12/24-R	Strobe Horn
	MT-24-LS-VFR	Strobe Horn
	MT-24-LSM-VFR	Strobe Horn
	MT-24-MS-VFR	Strobe Horn
	MT-24-IS-VFR	Strobe Horn
	MT-24-SL-VFR	Strobe Horn
	MT-24-SLM-VFR	Synch. Multitone Strobe
	MT-24-WM	Strobe
	MT-24-WM-VF-R	Horn
	MT-24-WM-VFR	Horn
	RS-2415-HFR	Strobe
	RSP-2415-VFR	Strobe
	RS-241575-VFR	Strobe
	RSP-241575-VFR	Strobe
	RS-2430-VFR	Strobe
Wheelock (cont.)	RS-2430-HFR	Strobe
Wheelock (cont.)	RS-2475-VFR	Strobe
	RSP-2475-HFR	Strobe
	RS-24110-HFR	Strobe
	RSP-24110-HFR	Strobe
	SL-24-VFR	Synchronized Remote Strobe
	SL1-24-VFR	Synchronized Remote Strobe
	SL3-24-VFR	Synchronized Remote Strobe
	SLP-24-VFR	Synchronized Remote Strobe
	SLM-24-VFR	Synchronized Remote Strobe
	SL1M-24-VFR	Synchronized Remote Strobe
	SL3M-24-VFR	Synchronized Remote Strobe
	SLPM-24-VFR	Synchronized Remote Strobe
	SHW-24-VFR	Synchronized Remote Strobe
	SH2W-24-VFR	Synchronized Remote Strobe
	SHPW-24-VFR	Synchronized Remote Strobe
	SCM-24-R	Controller for Synchronized Strobes
	SM-12/24-R	Sync Module
	SR-2415-VFR	Sync Strobe

Table 6-7:	Compatible	24-Volt	Notification	Devices
	oompanbic		Notifioation	Devideo

Manufacturer	Model Number	Device Type
	SRP-2415-HFR	Sync Strobe
	SR-241575-VFR	Sync Strobe
	SRP-241575-VFR	Sync Strobe
	SR-2475-VFR	Sync Strobe
	SR-2475-HFR	Sync Strobe
	SR-24110-HFR	Sync Strobe
Wheelock (cont.)	SRP-24110-HFR	Sync Strobe
	V7001T-12\24-W-FR	Strobe Horn
	WM3T-24-FR	Remote Strobe
	WM3T-24-VFR	Remote Strobe
	WS1T-24-FR	Strobe
	WS3T-24-FR	Strobe
	WST-24-FR	Strobe

Table 6-7: Compatible 24-Volt Notification Devices

6.4 Auxiliary Relays

The 5204 provides two auxiliary relay outputs. One relay output annunciates alarms. The other can be programmed to annunciate either alarm or trouble conditions, or can be used to activate the Model 5220 Direct Connect Module. Figure 6-11 shows the relay contact connections.) The relays can be programmed to activate for the conditions below, either for all zones or by individual zones:

- Pre-alarm (entry delay) (Not acceptable for NFPA 72 central station)
- Fire alarm
- Auxiliary alarm
- Alarm by zone
- System or loop troubles (loss of AC power, low battery power, failure of 5205 to communicate, phone line troubles, and bell troubles)
- The Model 5220 Direct Connect Module (see Section 6.2.2)



Figure 6-11 Auxiliary Relays

6.5 External Silence Keyswitch (Optional)

For manual silencing or resetting alarms, you can attach a remote keyswitch to the 5204 at terminal #6. Use a UL listed keyswitch. The keyswitch will operate as Normally Open Momentary at 24 VDC/.25 A minimum.

Once the keyswitch has been wired, it must be programmed either to silence or reset alarms (see Section 8.4.1.). If programmed to silence, the keyswitch turns off an annunciator that is signaling a trouble or alarm condition.

If programmed to reset alarms, the keyswitch removes smoke detector power for a programmed length of time (see Section 8.4.1). This allows the smoke detector to sense new alarm conditions.



Figure 6-12 Wiring an External Silence or Reset Alarms Keyswitch

The external keyswitch silences notification devices and horns only, not the onboard beeper or local annunciators. See Section 7.1.3 if you need additional information.

If using a pushbutton reset, it must be placed within a firefighter's lockbox.

Section 7 Normal Operation

The optional Model 5230 Remote Annunciator provides annunciation of trouble and alarm conditions, and can be used to program the system. Key functions for both the Model 5230 (Figure 7-1) and the 5204 built-in touchpad (Figure 7-2) are described in Section 7.1. Section 8 explains how to program the 5204 using the Model 5230.



Figure 7-1 Model 5230 Remote Annunciator

The Model 5230 Remote Annunciator has a liquid crystal display (LCD) for displaying English-language messages. If the 5204 is not being programmed, the LCD cycles through all messages that are applicable at the time, showing a different one every 1.5 seconds. Refer to Section 9.3 for more information on troubleshooting messages.

When AC power is being supplied, and the battery is fully charged, the POWER LED glows steadily. If the POWER LED is flashing, the AC power has been removed or the backup battery is low. If neither AC nor battery power is being supplied, the POWER LED is off.

The audio transducer buzzer produces short beeps to annunciate keystrokes. It also emits a long, high-pitched tone to denote a trouble condition or to indicate that an annunciator function has been entered incorrectly (see Section 7.1).



Figure 7-2 Built-in Touchpad

7.1 Built-in Touchpad and Model 5230 Annunciator Operation

To operate the 5204, you must use either the built-in touchpad or the Model 5230 Remote Annunciator. This annunciator functions the same as the internal touchpad except for the STEP key. The installer uses this key to step through programming options (see Section 8.3).

Following are the basic operating functions. Note that if no keys are pressed for 15 minutes while in program mode, the system will time out and resume normal operation.

The message TRY AGAIN appears on the 5230 display if you do not press any keys for 5 seconds while accessing a function, or, if you attempt to access a function before exiting from another function.

In the following table, Code 0 refers to the installer's code (factory programmed as 5204). Code 1 refers to the operator's code (factory programmed as 1111). These two codes are described in Step 45 and Step 46 of Step Programming (see Section 8.4.1).

Note: A valid operating code is always required when using the 5230.

To:	Press:		Additional Information
10.	5230 Annunciator	Built-in Touchpad	Additional information
Clear	CLEAR	CLEAR	Enables you to start again if you enter the wrong keystrokes. If you enter a function incorrectly on the 5230, the annunciator's PZT buzzer will emit a long, high- pitched tone.
Test the system	TEST ENTER + code 0 or 1		The system will test the 4180 outputs, the built-in touchpad LED display, signaling devices, sirens, and communicator.
Reset alarms (or smoke detectors)	ENTER + code 0 or 1	RESET ALARM 1 + code 0 or 1	After a smoke alarm has been triggered, this function removes smoke detector power for the programmed length of time (as determined by the smoke detector). This allows the smoke detector to sense new alarm conditions.
When a trouble com memory. If you do implying incorrectl	dition occurs and you reset the alar not clear the alarm memory, the tro y that more than one trouble condit	m, the trouble condition is stored i ouble condition is displayed the ne tion exists.	n memory until you <u>clear</u> the alarm ext time a trouble condition occurs,
Clear alarm memory	$\underbrace{\begin{bmatrix} \text{CLEAR} \\ \text{MEMORY} \\ 2 \end{bmatrix}} \text{ENTER} + code \ 0 \text{ or } 1$		Clears all data out of alarm memory and resets the 4180. (This function removes all memory of alarms.)
Reset the dialer	$\overbrace{3}^{\text{DIAL.}} \textbf{ENTER} + code \ 0$	$ \underbrace{ \begin{bmatrix} \text{DIAL.} \\ \text{RESET} \\ \textbf{3} \end{bmatrix} }_{\textbf{S}} \textbf{ENTER} + code \ 0 \\ \\ \end{array} $	Aborts an in-progress call to the central station.
Initiate download	$ \begin{bmatrix} LOAD \\ 4 \end{bmatrix} \begin{bmatrix} ENTER \\ + code 0 \end{bmatrix} $	$ \begin{array}{ c c } \hline LOAD \\ 4 \\ \hline 4 \\ \hline + code 0 \end{array} $	Starts the downloading process. Exit the DOWNLOADING mode by pressing CLEAR CLEAR.
Display alarm memory	$\underbrace{ \begin{smallmatrix} \text{Display} \\ \text{MEMORY} \\ \textbf{5} \end{smallmatrix} } \textbf{ENTER} + code \ 0 \ or \ 1$	DISPLAY MEMORY 5	Displays the current alarm memory. (It is recommended that you clear alarm memory after displaying it.)
Display troubles	$\begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \\ \begin{array}{c} \\ \\ \\ \\ \\ \end{array} \end{array} \end{array} \end{array} \end{array} = \begin{array}{c} \begin{array}{c} \\ \end{array} \\ \begin{array}{c} \\ \end{array} \end{array} \\ \begin{array}{c} \\ \end{array} \\ \end{array} = \begin{array}{c} \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \end{array} \\ + \begin{array}{c} \\ \\ \end{array} \\ code \ 0 \ or \ 1 \end{array} \end{array}$	DISPLAY TRBL. 6	Displays trouble conditions.
Silence trouble or alarm condi- tions	$\underbrace{\frac{\text{SILENCE}}{\text{STEP}}}_{+ \text{ code } 0 \text{ or } 1$	SILENCE	Silences signaling devices that are in trouble or alarm. (On- board beeper and local annun- ciators continue to sound until serviced. See Section 9.1 for more details.)

Table 7-1: Touchpad Operations

То	Press:		Additional Information		
10.	5230 Annunciator	Built-in Touchpad			
Fire drill	$\begin{array}{c} \begin{array}{c} \begin{array}{c} \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\$	$\underbrace{\begin{smallmatrix} \text{CLEAR} \\ \text{MEMORY} \\ \textbf{2} \end{smallmatrix}}_{\textbf{2}} \underbrace{\texttt{TEST}}_{\textbf{0}} \underbrace{\texttt{ENTER}}_{\textbf{+} \operatorname{code} 0$ To end a fire drill:	Causes the system to sound an alarm and report a FIRE TEST.		
	To end a fire drill: $\underbrace{SILENCE}_{STEP} + code 0 \text{ or } 1$	SILENCE			
Set time		SET TIME 9 ENTER	See explanation below.		
The SET MODE SET TIME mode second digit indio digit, the SET TI	The SET MODE LED will turn on and the built-in touchpad display will flash 9- indicating that you are in the SET TIME mode. You must enter six digits for the time. The first digit is the day of the week (see below). The second digit indicates time of day (see below). The last four digits are the actual time. Upon entering the sixth digit, the SET TIME LED will turn off indicating that you have set the time.				
Day of week: 0 =	Sun., 1= Mon., 2 = Tue., 3 = V	Ved., $4 =$ Thu., $5 =$ Fri, $6 =$ Sat			
Time of Day: 0 =	= AM, 1 = PM				
Example: To ente	er the time of Wed., 4:30 p.m.,	you would enter the following	digits: $3 + 1 + 0 + 4 + 3 + 0$.		
Note: The 5204 powers up in the SET TIME mode, with 9- showing on the display. If you wish to set the time at this point, it is not necessary to press the 9 ENTER (code) key sequence. Simply key in the appropriate six digits. To exit the SET TIME mode, press ENTER.					
Disable/Enable (shunting / un- shunting)	$(\text{Zone #1-4}) + \underbrace{\frac{\text{DISABLE}}{\text{SHIFT}}}_{\text{code 1 or 0}} +$	(Zone #1-4) + $(Disable) + code1 or 0$	Disables a zone (prevents it from responding to an alarm condition) or reactivates a dis- abled zone. When you disable, a trouble buzzer will sound.		
<i>Note:</i> If the dialer is busy, modes 22, 25, and 27 are disabled.					
If you a	re in mode 22, 25, or 27, the dia \sim	aler is disabled.			
Walk test	CLEAR MEMORY 2 CLEAR 2 ENTER + code 0 (factory programmed as 5204) + To exit press: SILENCE STEP SILENCE STEP CLEAR CLEAR	$\begin{array}{c} \hline \textbf{CLEAR} \\ \textbf{MEMORY} \\ \textbf{2} \\ \textbf{2} \\ \textbf{2} \\ \textbf{ENTER} \\ \textbf{+ code } 0 \\ (factory programmed as \\ 5204) \\ To exit press: \\ \hline \textbf{SILENCE} \\ \hline \textbf{SILENCE} \\ \hline \textbf{SILENCE} \\ \hline \textbf{CLEAR} \\ \hline \textbf{CLEAR} \\ \hline \textbf{CLEAR} \\ . \\ \end{array}$	Enables you to test the system. When you enter this mode, the LCD will indicate that you are in the Walk Test mode. When a zone is violated, the signaling device outputs will become ac- tive for approximately 6 sec- onds.		

Table 7-1: Touchpad Operations



Table 7-1: Touchpad Operations

7.1.1 Operating Modes

The following table describes which codes can access operating modes during alarms:

Operating Mode		Allowed	Code Required	
		Alarm	On 5230	On Built-in Touchpad
00	System test	No	Code 0 or 1	None
01	Reset alarm	Yes	Code 0 or 1	Code 0 or 1
02	Clear alarm memory	No	Code 0 or 1	None
03	Dialer reset	Yes	Code 0	Code 0
04	Download	No	Code 0	Code 0
05	Display alarm memory	No	Code 0 or 1	None
06	Display troubles	No	Code 0 or 1	None
09	Set time	No	Code 0 or 1	Code 0 or 1
2B	Silence mode	Yes	Code 0 or 1	None
20	Fire drill	No	Code 0 or 1	Code 0 or 1
22	Walk test	No	Code 0	Code 0
27	Program	No	Code 0	Code 0
25	Troubleshooting	No	Code 0	Code 0
E0	Disable/enable zone	Yes	Code 0 or 1	Code 0 or 1

7.1.2 Built-in Touchpad Display Codes

The built-in touchpad display shows the zones in which a trouble or alarm condition is occurring. It also displays two-digit codes that represent a variety of conditions, as an aid in troubleshooting the system. These codes are listed below.

The following table describes the codes that appear on the built-in touchpad:

Display	Explanation
0	Fire drill (with ALARM, ALARM MEMORY, or TROUBLE LED).
1 through 4	Zone numbers (with ALARM, ALARM MEMORY, or TROUBLE LED). A "c" in front of the number indicates a supervisory sprinkler zone.
E0 E7	Indicates trouble with the dialer microprocessor. Indicates trouble with the EEPROM memory.
F0 F1 through F7	5230 annunciator power trouble. Indicates trouble with a particular annunciator.
A1 through A2	Indicates trouble with a particular bell output.
P1	Indicates trouble with the smoke detector power.
A1 through A4	Indicates trouble with the accessory power (terminal 26).
Р3	P3 indicates a short between Earth Ground and Common Ground. To determine the location of the short, remove field wiring circuits until the control returns to normal operation. When the circuit that caused the trouble is found, use an ohmmeter to measure the resistance between each wire in the circuit and earth ground terminal #52. The resistance must be higher than 100K ohms.
P4	P4 indicates a short between Earth Ground and loop or bell power. To determine the location of the short, remove field wiring circuits until the control returns to normal operation. When the circuit that caused the trouble is found, use an ohmmeter to measure the resistance between each wire in the circuit and earth ground terminal #52. The resistance must be higher than 500K ohms.
PO	Indicates that the printer is out of paper.
dC dF dL	Low battery condition. Low AC condition. Data lost during an attempt to transmit data to the central station.
L1 L2	Phone Line 1 Fault Phone Line 2 Fault
-0 -2 -4 -5 -6 -7 -8 -9	Fire drill Walk test Downloading Zone test HEX PROGRAMMING mode STEP PROGRAMMING mode SET DATE mode SET TIME mode
2-, 3-, 4-, 5-, 6-, 7-, 8-, 9-	User must enter a code to perform the desired function with these prompts.

7.1.3 Silencing the System

To silence a trouble, press SILENCE.

To silence an alarm, follow these steps:

1. Disable the zone by pressing (zone number) + DISABLE + code 1 or 0.

2. Reset the system by pressing $\begin{bmatrix} \text{RESET} \\ 1 \end{bmatrix}$ ENTER + code 1 or 0.

3. The zone is now in trouble because of the disabled zone and can be silenced in the normal way by pressing SILENCE.

See Section 9.1 for related information.

7.1.4 LED Indicators

Six light emitting diodes (LEDs) appear in the 5204 cabinet window.

LED	Status	Condition
	OFF	No alarm condition exists.
ALARM (red)	ON	A fire alarm condition exists in the zones shown on the touchpad.
	OFF	An alarm or trouble has not been silenced.
SILENCED (yellow)	ON	An alarm or trouble condition exists and the audible annunciators have been silenced.
AC / DC (green)	OFF	Panel has lost all power.
	ON	Panel is running on AC and battery power (normal condition).
	FLASHING	Panel is running on battery power only or AC power only.
MEMORY (vellow)	OFF	No information is stored in alarm memory.
MEMORT (yenow)	ON	An alarm condition has been reset.
TROUBLE (yellow)	OFF	No trouble condition exists.
	ON	A trouble condition exists.
<u>SET MODE</u> (yellow) REPORT	OFF	Normal operating mode and not reporting.
	ON	System is in a SET (TEST or PROGRAM) mode.
	FLASHING	System is reporting

7.2 System Testing

System testing includes fire drills, zone testing, and 24-hour automatic tests.

7.2.1 Fire Drills (Mode 20)

You can run fire drills using either the built-in touchpad or the Model 5230 touchpad. To

initiate a fire drill, press $\underbrace{\begin{bmatrix} CERR\\ 2 \end{bmatrix}}_{2} \underbrace{\begin{bmatrix} TEST\\ 0 \end{bmatrix}}_{2}$ ENTER + code 0 or 1. The system will sound an alarm and report a fire test. To end the fire drill, press SILENCE) + code 0 or 1.

7.2.2 Walk Test (Mode 22)

The Walk Test mode enables you to test individual sensors.

Zones can be disabled individually to facilitate testing and troubleshooting. Disabled zones will NOT be tested. If no zones are tripped during the Walk Test (or keys pressed) for 15 minutes, the system will time out and resume normal operation.

To exit Walk Test mode, press STEP STEP CLEAR CLEAR. If using the built-in touchpad, press SILENCE SILENCE CLEAR CLEAR.

7.2.3 Automatic Self Test

The Model 5204 lets you select the time of day to send the 24-hour automatic test signal to the central station.

The Auto Test (Dialer test sent automatically at specified times) also sends all unrestored events, as now required by UL. Events listed before AUTO TEST on the printout at the central station are new events. Events listed after AUTO TEST are old events that have not been restored.

7.2.4 Watchdog Circuit

During normal operation, the control microprocessor of the 5204 is constantly running programs to check inputs and carry out other routine functions. If this program stops running for some reason, the watchdog circuit will automatically attempt to resume normal operation by resetting the microprocessors. Each time the watchdog circuit initiates a reset signal, it will also sound the audible trouble signal for approximately four seconds.

7.3 Zone Characteristics

This section describes the programmed characteristics of zone inputs. Zone characteristics include zone type, cross alarm, pre-alarm, smoke verification, and zone response time.

7.3.1 Zone Type

If the 5204 should supervise a sprinkler system, select the zone as "sprinkler" even if dialer/ reporting has not been selected.

Undefined zones may be used to supervise the functioning of a commercial process (such as manufacturing operations, temperature control for heating or refrigerating systems, and so on), when failure of the supervised process could result in fire or explosion endangering life or property. Audible indicators for Fire and Waterflow zones override audibles for undefined zones.

If you are using waterflow switches with built-in delays, note that zone response time must not exceed 120 seconds.

7.3.2 Cross Alarm

Cross alarm areas require a minimum of 2 detectors on 2 different zones.

Do NOT select this feature if smoke verification has been selected in Steps 25-28.

If a zone is programmed for cross alarm and an alarm condition occurs in this zone, relay 1 will be activated. An alarm sound or report will not be generated until an alarm condition also occurs in one of the other zones.

If an alarm has already been activated on the other zone, and an alarm subsequently occurs in a cross-alarm zone, the cross-alarm programming will be ignored. The alarm condition in the cross-alarm zone will immediately generate an alarm sound and report.

Example

In the following example, various combinations of alarm conditions are shown and whether an alarm sound or report is generated.

Alarm Condition Exists in Zones	Alarm Generated?	Alarm Generated?
	(Cross Alarm #1 Selected)	(Cross Alarm #1 and #3 Selected)
1	No	No
1,2	Yes	Yes
1,2,3	Yes	Yes
1,2,3,4	Yes	Yes
2	Yes	Yes
2,3	Yes	Yes
2,3,4	Yes	Yes
3	Yes	No
3,4	Yes	Yes
4	Yes	Yes

Table 7-2

When using cross alarm, be sure to place smoke detectors so that no two adjacent detectors



belong to the same zone. Refer to the following diagram:

Distribution of detectors for cross alarm in a large open area with 4 zones. For a smaller installation, use any segment of this example

If you are installing an NFPA 72 Protected Fire Alarm system, do not use the cross alarm and alarm verification features in the same installation (you can use one or the other, but not both).

Local Protected Fire Alarm systems that require the activation of two smoke detectors to produce the alarm response shall be permitted, providing:

- 1. They are not prohibited by the authority having jurisdiction.
- 2. There are at least two detectors using two different zones in each protected space.
- 3. Detector spacing is no more than one half that determined by the application of NFPA 72, *Standard on Automatic Fire Detectors (NFPA National Fire Alarm Code, 1993 Edition, Chapter 5).*
- 4. The alarm verification feature is not used.

7.3.3 Pre-Alarm

This option programs the number of seconds (5-45) on alarm will be delayed. See description of "pre-alarm" in Steps 19-22.

Alarm bells and reporting will be delayed in zones 1-4 (respectively). Gives an audible tone that the system is about to go into alarm (Yes or No). Can be used to allow for immediate evacuation. (Duration of delay programmed in Step 18.)

7.3.4 Smoke Verification

In UL installations, the total delay, including detector built-in start-up time and programmed reset time, must not exceed 60 seconds. To help you determine the maximum length you can select for smoke verification time, use the following calculation:

60 - [detector start-up time + detector reset time] = smoke ver. time

If you are installing NFPA 72 Local Protected Fire Alarm system, do not use the cross alarm and alarm verification features in the same installation (you can use one or the other, but not both).

Refer to your smoke detector documentation for start-up and reset times. Reset times are also listed in Table 6-2 and Table 6-3.

Do NOT select this feature if cross alarm has been selected in Step 15-Step 18.

Figure 7-3 shows the timing of the smoke verification process:



Figure 7-3 Smoke Verification Process

7.3.5 Zone Response Time

The zone (loop) response time for trouble conditions is 3 to 4 seconds.

Requirements/Restrictions

- The delayed responses (speeds 1-3) are NOT intended to be an alarm verification feature.
- Delays may be used ONLY on waterflow switches. The zone response time must not exceed 120 seconds for waterflow switches that have their own delay.
- Class B (style B) zones must be programmed as "1".
- Do NOT enter any numbers other than 0-3. Doing so will cause the 5204 to default to one of these four speeds.

Section 8 Programming

The Model 5204 system can be programmed for specific customer needs through the use of programming options.

Using Step Programming or downloading, you can reprogram the options that are stored on an Electrically Erasable Programmable Read Only Memory (EEPROM) chip. Step Programming (Section 8.3) is available through the Model 5230 Remote Annunciator and the built-in touchpad.

If your system includes the Model 5205 Dialer Module, you can also use the Model 5541 Downloading software (Section 8.2).

This section explains programming only; refer to Section 7 for basic operating instructions.

8.1 **EEPROM Information**

The EEPROM is used to store all of the system options, such as system configuration, telephone numbers, reporting formats, account numbers, and so on. The EEPROM is an 8-pin integrated circuit chip that can be reprogrammed up to 1000 times. It will retain programmed information even after a total loss of power. Default values for all options are preprogrammed into the EEPROM before it leaves the factory. These default values are listed in the Quick Reference table in Appendix A.

Caution

Do NOT attempt to use the step programming feature with a blank EEPROM. You MUST use an EEPROM that contains the default data or one that has been previously programmed. The 5204 is shipped from the factory with an EEPROM programmed with default data.

To order EEPROMs, call Silent Knight Sales at (800) 446-6444. The number to use when ordering is 009355.

8.2 Downloading

The Model 5541 Downloading software (Revision 3.33 or higher) enables you to use a computer at a remote location to reprogram options at a particular installation.

Note: Revision 3.7 (or later) of the downloading software is required if you wish to report sprinkler supervisory event codes to the central station. See Step 56-Step 61 for more details.

The downloading software is organized into menus. As you move through the menus, the screens tell you how to select options. The programming options are described in detail in Section 8.4. For complete information on using the software, see the Model 5541 Downloading Software Manual (P/N 150497).

After the 5204 accepts all the new data, the 5204 will transmit the message PROGRAMMING PASS in SIA format to the 9000 receiver. UL requires the central station to send a representative to the site to verify the programming changes. (If the 5204 does NOT accept all the new data, the 9000 will print the message, PROGRAMMING FAIL.)

To be able to print PROGRAMMING PASS or PROGRAMMING FAIL, the 9000 must have Model 9307 software package, Revision 900501 or later.

This software can be used only with the Model 9200-E CPU card. (If you need information about how to contact Silent Knight for an upgrade, see Section 8.1.)

8.3 How to Use Step Programming

You can access Step Programming (also known as mode 27) using either the Model 5230 Remote Annunciator or the built-in touchpad to program options. The Step Programming form in this section describes the options and how to program using either method.

If your 5204 installation does not include the Model 5230, it is possible to connect a temporary annunciator for programming on the 5204 circuit board. Refer to Section 5.8 for more details.

If no key is pressed for four minutes while in programming mode, the system will exit programming mode.

When you key in a function very quickly, wait for the message to appear on the display before you press the ENTER key.

8.3.1 Entering Step Programming (Mode 27)

- 1. Read through the options in this section to determine what selections you wish to make for each option. For future reference, you can write down your selections in the Quick Reference table in Appendix A.
- 2. Enter the Step Programming mode by pressing 2 Dr ENTER + code 0 (factoryprogrammed as 5204). If you have entered mode 27 correctly, the following information will appear in the display:

Model 5230	Built-in Touchpad
The first line of the LCD will show the programming option for step 1. The second line will show the most recently programmed value for that option. For example: 24–U SMOKE POWER YES	Depending on whether step 1 is programmed as Yes or No, the display will show either of the following:

If you get a trouble beep and the message TRY AGAIN appears, either you are not using the correct code 0, or the EEPROM could be malfunctioning. To correct this problem, you must obtain a new default EEPROM (see Section 8.1 for ordering details).

8.3.2 **Programming Options**

For selecting options, entering digits works as follows:

Scroll For most options, you enter numbers in the same way as if you were using a calculator. The digits appear on the right side of the display and scroll to the left as you continue to enter data.

To program an option, key in the data you have written on the Quick Reference table in Appendix A for that option.

Model 5230	Built-in Touchpad
The second line of the LCD will show the new value. For example, if you changed the default setting of Yes for Step 1 to No, the display would appear as follows: 24-U SMOKE POWER NO Press ENTER . The display will advance to the next step. To select Yes or No Press any digit to toggle the Yes/No option.	The display will show the first digit of the new data. Press ENTER . The display will show the first digit of the next step. If the data includes more than two digits, the left most digit will shift off the display each time you key in a new digit. For example: Suppose you want to program ACCOUNT #1 (Step 66) as 123456. After you key in RESET (Step 66) as 123456. After you key in (Step 66) as 123456. After you key

8.3.3 Advancing to the Next Option

To skip a step, press ENTER. The data in the skipped step will not change.

Model 5230	Built-in Touchpad
The LCD will show the next option.	The display will show the previously programmed data for the next step.

8.3.4 Going to a Specific Step

Model 5230	Built-in Touchpad
The LCD will show the option name. Press ENTER to continue programming that step.	Press SLENCE . The display will show the step number and the ALARM LED will light. The display indicates the current step number. Press ENTER to continue programming that step.
Press STEP) to go to a specific step. The first line of the LCD will show ENTER THE STEP # and the current step number. Key in the new step number, followed by ENTER . The LCD will show the new option name.	To go to a specific step, enter the new step number. The display will show the previously programmed data for the new step, and you can now continue programming this step.
Note: If you try to go to a step that does not exist, the display will go back to the previous step.	Note: If you try to go to a step that does not exist, the display will go back to the previous step.

8.3.5 Viewing Previously Programmed Data

To view the previously programmed data for a particular step (in other words, you have not entered any new data), go to the step as explained in Section 8.3.4.

Model 5230	Built-in Touchpad
The data for that step will show on the second line of the LCD.	The display will show the first digit. If the data includes more than one digit, press \bigcirc LEAR to view the next two digits. Press \bigcirc as often as necessary until you have viewed all the digits.

8.3.6 Correcting Errors

Press the CLEAR key to erase the step number or when you make a mistake when entering data. If you have not yet touched the ENTER key, press the CLEAR key. Key in the correct data and press ENTER.

8.3.7 Entering Hexadecimal Digits

To enter numbers greater than 9, use the SHIFT key as shown below to enter numbers 10-15. Hexadecimal digits (in parentheses) appear on the screen to represent these numbers:

From The 5230	From the Built-In Touchpad
$\begin{array}{ c c }\hline \hline \textbf{DISABLE}\\ \hline \textbf{SHIFT} \end{array} \begin{array}{ c c } \hline \textbf{RESET}\\ \textbf{ALARM}\\ \textbf{1} \end{array} = 10 \ (A) \end{array}$	DISABLE $\begin{array}{c} \textbf{LOAD} \\ \textbf{4} \end{array} = 13 \text{ (D)}$
$\overbrace{\begin{array}{c} \textbf{DISABLE}\\ \textbf{SHIFT} \end{array}}^{\text{CLEAR}} 2 = 11 \text{ (B)}$	$Disable \begin{bmatrix} Display \\ MEMORY \\ 5 \end{bmatrix} = 14 (E)$
$\begin{array}{ c c }\hline \hline \textbf{DISABLE}\\ \hline \textbf{SHIFT}\\ \hline $	$bisable \begin{bmatrix} bisplay \\ TRBL. \\ 6 \end{bmatrix} = 15 \text{ (F)}$

8.3.8 Programming Examples

The following examples may help you to understand how to use Step Programming. The selections you make in each installation will vary depending on each customer's needs. The way you move through Step Programming may also vary from how it is described here.

The following examples assume you are at the Step 1 display (see Section 8.4.1).

Example 1: Choosing a Programming Option from a Menu

Suppose you want to program the 5204 so that Bell #2 will pulse in Zone #3.

- 1. Press STEP $\begin{bmatrix} LOAD \\ 4 \end{bmatrix} \xrightarrow{CLEAR \\ MEMORY \\ 2 \end{bmatrix}$ ENTER to go to Step 42.
- 2. The following display will be shown:

<u>Mc</u>	odel 5230	<u>Touchpad</u>
BELL#2 : STEADY	ZONE #2	٥

- 3. Press $\begin{pmatrix} \text{RESET} \\ \text{ALARM} \\ 1 \end{pmatrix}$ for PULSE.
- 4. Press ENTER).

Example 2: Programming Location Description Names

Suppose you want to program the Model 5230 Annunciator to display meaningful location names for Zones 2 and 4. The words you wish to display are GARAGE for Zone 2 and EAST OFFICE for Zone 4. You can select these words from the 5230 library of names using Step Programming. See Table 8-2 for a complete list of words contained in the library.

- Programming a One-Word Display
- 1. In programming mode, press STEP $\begin{pmatrix} \text{RESET}\\ 1 \end{pmatrix} \begin{bmatrix} \text{CLEAR}\\ 2 \end{bmatrix}$ ENTER to go to Step 12.
- 2. Press $\begin{bmatrix} \text{RESE} \\ 1 \end{bmatrix}$ repeatedly until the word GARAGE displays on the LCD. Press ENTER to select.
- Programming a Two-Word Display

To program Zone 4 to display EAST OFFICE, you will have an additional step since you are programming two words instead of one.

1. In programming mode, press $\begin{bmatrix} RESET \\ ALARM \\ 1 \end{bmatrix} \begin{bmatrix} LOAD \\ 4 \end{bmatrix}$ ENTER to go to Step 14.

- 2. Press $\begin{bmatrix} RESET \\ ALARM \\ 1 \end{bmatrix}$ repeatedly until the word EAST displays on the LCD.
- 3. To add the second word, press $\underbrace{\begin{bmatrix} CLEAR\\ MEMORY\\ 2 \end{bmatrix}}_{2}$ until you reach OFFICE. Press ENTER to select.

8.3.9 Exiting Step Programming

At any time while using Step Programming, you can exit mode 27 by pressing:

From The 5230	From The Built-in Touchpad	
SILENCE STEP STEP CLEAR CLEAR	SILENCE SILENCE CLEAR CLEAR	

8.4 Step Programming Options

This section explains the options you can program for the Model 5204 panel. Before you begin step programming, read through this and write down the selections you wish to make on the perforated form provided in Appendix A.

For each option, the values programmed at the factory (defaults) are shown in the sample displays. The valid input selections are in parentheses in the Option Descriptions column. If you do not reprogram a particular option, the default values will be in effect.

These options are listed in the order in which they appear on the 5230 annunciator. When you are using the built-in touchpad or the Model 5541 downloading software menus, the order and the option names vary slightly.

8.4.1 Programming Steps

Table 8-1 lists all the steps numbers, what is displayed, and the choices available in those steps.

	Displayed Information			
Step #	On the 5230	On the Built-In Touchpad	Description	
Step 1	24-V SMOKE POWER	Ч	Select this option if you want the 5204 to provide 24 V of smoke power. If this option is not selected, the 5204 will provide 12 V of smoke power. (Yes or No) Y = 24 V panel N = 12 V panel	
Step 2	EXTERNAL SILENCE	Ч	This option controls the external input that can be used to silence audible alarms or to reset alarms (Yes or No). If an external keyswitch has been wired, Yes = Silences alarms No = Resets alarms	
Step 3	#SUPER ANNUNC	0	This number (0-3) specifies the number of annunciators that will be supervised. If 0 is selected, it is possible to use all three annunciators, but none will be supervised.	
Step 4	SMOKE RESET TIME	P	Enter (2-7 seconds) the length of time power is removed from the smoke detector after it has been reset (see Section 7.1). Note: Manually silencing the annunciators by pressing SI- LENCE) (code 1 or 0), does NOT reset the smoke detector. To remove and return power, so the smoke detector can continue to sense alarm conditions, press 1)ENTER). See the tables in Section 6.1 for the smoke reset time required for each smoke detector model.	
Step 5	ZONE DISABLE	Ч	Selections are the following: Yes = All zones can be disabled. No = Zones cannot be disabled.	

Table 8-1: List Of Programming Steps

Table	8-1: L	ist Of	Programming	Steps
-------	--------	--------	-------------	-------

	Displayed Information			
Step #	On the 5230	On the Built-In Touchpad	Description	
Step 6	LATCH SPRINKLER	no	Set how the sprinkler zone will operate. Press any numeric- digit to toggle the selection from Yes to No. Yes = When the sprinkler zone shorts for a duration longer than the Zone Response (set in steps 30 through 33), the annunciator remains active until reported or manually silenced. No = When the sprinkler zone shorts for a duration longer than the Zone Response (set in steps 30 through 33), the zone will follow system status and indicate a supervisory on that zone for the duration of the faulted condition.	
Step 7	ZONE TYPE:#1		Selects the zone type for zones 1-4 respectively. This is the	
Step 8	ZONE TYPE:#1		the Model 5205 and a SIA reporting format is used. Possible	
Step 9	ZONE TYPE:#1		zone types are the following:	
Step 10	ZONE TYPE:#1		0 = Fire 1 = Waterflow 2 = Undefined 3 = Sprinkler	
Step 11	LOCATION ZONE#1		Enter a two-word location description for zones 1-4. Select the words from the list below. If you are programming with the built-in touchpad, use the SHIFT key to advance to the corresponding two-digit number.	
Step 12	LOCATION ZONE#2		Press 1) as many times as necessary to advance through the word list for the first word. (Press 6) to go backwards through the word list to choose the first word.)	
Step 13	LOCATION ZONE#3	[Blank]	Press 2) as many times as necessary to advance through the word list for the first word. (Press 7) to go backwards through the word list to choose the first word.)	
Step 14	LOCATION ZONE#4		 OR Press SHIFT) Press 1) or 2) (for first or second word) Press digit for number of word from word list. See Table 8-2 for complete word list. 	
Step 15	CROSS ALARM#1		If this option is selected,	
Step 16	CROSS ALARM#2	_	an alarm condition exists in another zone.	
Step 17	CROSS ALARM#3	no	No = Alarm in zone sounded or reported immediately.	
Step 18	CROSS ALARM#4		Cross alarm areas require a minimum of 2 detectors on 2 different zones.	
Step 19	PRE-ALARM TIME	5	This option programs the number of seconds (5-45) on alarm will be delayed. See description of "pre-alarm" in Step 20-Step 23.	

	Displayed Information			
Step #	On the 5230	On the Built-In Touchpad	Description	
Step 20	PRE-ALARM #1		Alarm bells and reporting will be delayed in zones 1-4	
Step 21	PRE-ALARM #2		(respectively). Gives an audible tone that the system is about to go into alarm (Yes or No). Can be used to allow for	
Step 22	PRE-ALARM #3	no	immediate evacuation. (Duration of delay programmed in Step	
Step 23	PRE-ALARM #4		19.) See Section 7.3.3 for more details.	
Step 24	SMOKE VER TIME	18	Enter the duration of the smoke detector delay (5-18 seconds). See Section 7.3.4 for more details.	
Step 25	SOUND SMOKE VERIFY	no	If this option is selected, there will be an audible trouble signal whenever a smoke detector enters its verification period before causing an alarm (Yes or No). The duration of this delay is programmed in Step 24. See Section 7.3.4 for more details.	
Step 26	SMOKE VERIFY #1		If you select this option, zones 1-4 respectively will be	
Step 27	SMOKE VERIFY #2	_	controlled by the smoke detector delay that was programmed in Step 24 (Yes or No).	
Step 28	SMOKE VERIFY #3	no	Do NOT select this feature if cross alarm has been selected in	
Step 29	SMOKE VERIFY #4	_	Step 15-Step 18. See Section 7.3.4 for more details.	
Step 30	ZONE RESPONSE#1		Select the speed at which zones #1-4 will respond to alarm	
Step 31	ZONE RESPONSE#2		conditions (0-3). The following speeds are associated with each selection:	
Step 32	ZONE RESPONSE#3		0 = 0.3 to 0.4 sec.	
Step 33	ZONE RESPONSE#4	0	1 = 3 to 4 sec. 2 = 15 to 20 sec. 3 = 30 to 40 sec. The zone (loop) response time for trouble conditions is 3 to 4 seconds. See Section 7.3.5 for more details.	
			Controls what relay #2 will activate (0-3).	
Step 34	RELAY #2	0	0 = Relay #2 activates ALARM. 1 = Relay #2 activates TROUBLE. 2 = Relay #2 activates 5220 Direct Connect Module. 3 = Relay #2 activates 5220 City Box.	
Step 35	BELL #1 SILENCED	no	Select this option to enable notification device #1 to be silenced by pressing the SILENCE key (Yes or No).	
Step 36	BELL #1 ZONE #1		These options control the disposition of notification device #1	
Step 37	BELL #1 ZONE #2		in zones 1-4 respectively (0-4). 0 - Steady	
Step 38	BELL #1 ZONE #3		1 = Pulse (.5 seconds on, .5 seconds off)	
Step 39	BELL #1 ZONE #4		2 = Temporal (3.5 seconds on, .5 seconds off, 3.5 seconds on, .5 seconds off) 3 = Supervisory (1 second on, 2 seconds off) 4= ANSI Temporal 5 = Not used	

Table	8-1: Li	st Of P	rogramming	Steps
-------	---------	---------	------------	-------
	Displayed In	formation		
---------	------------------	-----------------------------	---	--
Step #	On the 5230	On the Built-In Touchpad	Description	
Step 40	BELL #2 SILENCED	0	Select this option to enable notification device #1 to be silenced by pressing the SILENCE key (Yes or No).	
Step 41	BELL #2 ZONE #1		These options control the disposition of notification device #1	
Step 42	BELL #2 ZONE #2		in zones 1-4 respectively (0-4). 0 = Steady	
Step 43	BELL #2 ZONE #3		1 = Pulse (5 seconds on 5 seconds off)	
		٥	2 = Temporal (3.5 seconds on, .5 seconds off, 3.5 seconds on, .5 seconds off)	
Step 44	BELL #2 ZONE #4		3 = Supervisory (1 second on, 2 seconds off) 4= ANSI Temporal	
			5 = Not used	
Step 45	INSTALLER'S CODE	5	The installer's code. This 4-digit code (also known as Code 0) is used by the installer to initiate downloads (see Section 7.1), set the time, and enter the programming mode on the Model 5230 Annunciator. It can also be used to perform the same functions as code 1. (Factory-programmed as 5204.)	
Step 46	OPERATOR'S CODE	J	The operator code. This 4-digit code (also known as Code 1) is used by the operator to silence annunciations and perform manual tests. (Factory-programmed as 1111.)	
Step 47	DIALER TYPE	٥	This option indicates the type of dialer used (0-3). 0 = Model 5205 Dialer NOT used 1 = USA 2 = 9000 Direct (do not select) 3 = European (used for Europe and Asia)	
Step 48	COMPUTER ENABLE	Ľ	If the downloading computer is to be used for programming and status request, you must select the COMPUTER ENABLE option. Press any number to toggle between Yes or No.	
Step 49	GROUND START	no	Select this option if you want to use ground start phone line instead of loop start (Yes or No). Yes (Ground Start); No (loop start). Do NOT select ground start in UL installations. The ground start option requires the optional Model 2608 relay. (See Section 5.7 Figure 5-3.)	
Step 50	TOUCHTONE LINE 1	no	This option determines whether the 5205 line #1 will use rotary dialing, or try both Touch-Tone and rotary dialing when trying	
Step 51	TOUCHTONE LINE 2	no	to send a report (Yes or No). Press any number to toggle between Yes or No.	

	Displayed In	formation		
Step #	On the 5230	On the Built-In Touchpad	Description	
			When the 5204 generates a report, it attempts to send the report to the priority central station phone number (see Steps 61-64). If the priority phone number is not available, the system tries the other phone number. It continues to alternate between the two phone numbers until the report is sent to one of the phone numbers.	
Step 52	MUST REPORT #1	Я	If account #2 is the first central station number available, the report will go to account #2. However, if the MUST REPORT #1 option has been selected, the system will continue to try to send the report to account #1, until account #1 is also available. Selections are:	
			Yes = Always send reports to central station account #1. No = If another phone # is available first, no report to account #1.	
Step 53	MUST REPORT #2	no	If account #1 is the first central station number available, the report will go to account #1. However, if the MUST REPORT #2 option has been selected, the system will continue to try to send the report to account #2, until it has either succeeded, or exhausted the programmed number of attempts (see Steps 67 and 72), leading to a dialer-failed condition. Selections are: Yes = Always send reports to central station account #2.	
			No = If another phone # is available first, no report to account #2.	
Step 54	AC LOSS HOURS	б	Using hexadecimal digits, select the number of hours that the AC power must be removed from the panel before the AC power loss is reported to the central station (6-15). If AC power is restored and lost again during this time period, the system will reset the time to 0 and start counting again. To enter hexadecimal numbers 10-15, Press SHIFT 1 -SHIFT 6. Do NOT select fewer than 6 or more than 12 hours for NFPA 71. For NFPA 72 Central Station Fire Alarm systems, set to 6-12 hours; for NFPA 72 Polarity Reversal, set to 15 hours.	
Step 55	# RINGS	10	If the downloading software (Model 5541) is to be used, this option determines the number of times the phone line will ring before the 5204 will answer the call. The allowable number of rings ranges from 2 to 14. If you select fewer than 2 rings, the 5204 will not answer. Use hexadecimal digits to program this step. If downloading is used, you must select the COMPUTER ENABLE option (Step 48).	

Table	8-1: Lis	t Of	Programming	Steps
-------	----------	------	-------------	-------

	Displayed In	formation			
Step #	On the 5230	On the Built-In Touchpad	Description		
Step 56	3∕1 ALARM CODE	1	Six types of events (alarm, sprinkler, trouble, disable, restore, and test) can be reported to the central station receiver. When transmitting this data using 3/1 format, each event is		
Step 57	3/1 SPRINKLER CODE	2	represented by a single digit. For each event, select a digit from 0-9 to identify that particular event to the receiver. The letters A (SHIFT 1) through E		
Step 58	3/1 TROUBLE CODE	8	(SHIFT 5) can also be used if the receiver is capable of receiving them.		
Step 59	3∕1 DISABLE CODE	5	 When using the 3/1 format, the receiver does not distinguish between 0 and A. 		
Step 60	3/1 RESTORE CODE	7	• When using the 3/1 format, much of the 5204's reporting ability is lost because you are limited to 1 digit to report an event. For example, the report does not indicate in which zone the event occurred		
Step 61	3∕1 TEST CODE	9	 If you are not using a 3/1 format, press the ENTER key as many times as necessary until you reach Step 62, ALARM #1 FIRST 		
			In Steps 61-64, you select the priority phone number for sending each type of report. That is, you select the phone number the 5204 will try to send the report first. Selections are: priority phone number will be #1 (Yes),		
Step 62	ALARM #1 1ST	Ч	or priority phone number will be #2 (No). If the priority phone number is not available, the 5204 will try to report to the other phone number. It continues to alternate phone numbers until is succeeds in sending the report to one of the phone numbers. If you want to be sure that the system will always report to a particular phone number (regardless of which number it reports to first), see Steps 52 and 53.		
Step 63	TROUBLE #1 1ST	Ч	Alarms and their restorals are reported on zones 1-4. The 5204 reports AC trouble, battery trouble, earth ground trouble, phone line #1 and #2 trouble, annunciator 1-3 trouble, and zone 1-4 trouble. Select this option if you want the system to try central station account #1 first when reporting troubles and trouble restorals. (See also Step 62.) Selections are: Yes = Try to report troubles to central station account #1 first. No = Try to report troubles to central station account #2 first.		
Step 64	DISABLE #1 1ST	Ч	Select this option if you want the system to try account line #1 first when reporting disabled zones. (See also Step 62.) Selections are: Yes = Report disabled zones to central station account #1 first. No = Report disabled zones to central station account #2 first.		

	Displayed Ir	nformation	Description		
Step #	On the 5230	On the Built-In Touchpad			
Step 65	TEST #1 1ST	Я	The 5204 reports manual and auto tests, and downloading successes and failures to the central station. Select this option if you want the system to try account line #1 first when reporting disabled zones. (See also Step 62.) Selections are: Yes = Try to report tests to central station account #1 first. No = Try to report tests to central station account #2 first.		
Step 66	ACCOUNT #1	1	Enter account number for central station phone #1 (6 digits; leading zeros if shorter). Factory default account number = 105204.		
Step 67	ATTEMPTS #1	Э	Enter the number of times the 5204 will try to dial each central station account # before the DIALER TROUBLE message appears on the 5230 Annunciator (3-5). Normally, the dialer will switch back and forth between account numbers after each attempt. If a different number of tries has been programmed on each number, the DIALER TROUBLE message will appear after all the attempts have been used up for the account number programmed with the lowest number of attempts. However, the dialer will continue to try to report on the remaining number until it has made as many attempts as have been programmed for that number.		
Step 68	FORMAT #1	5	Indicate the appropriate reporting format to be used on phone #1 (0-7). The numbers correspond to the formats described below. Refer to Section 10 for more information on reporting formats. 0 = SIA8. Security Industry Association standard. $1 = FSK81$. Silent Knight FSK format. Uses a 4-digit account number 2 = SK4+2. 20 pps pulsed-tone format. Uses a 4-digit account number. 3 = BFSK14. Format used with receivers that can receive BFSK and send a 1400 Hz acknowledgment tone. Uses a 3-digit account number. 4 = BFSK23. Format used with the Model 9000 receiver and other receivers that can receive BFSK and send a 2300 Hz acknowledgment tone. Uses a 3-digit account number. 5 = SIA20. Security Industry Association standard. 6 = 3/1 14 (6). Used with older Silent Knight, Ademco, or Sescoa receivers that send a 1400 Hz acknowledgment tone. The Model 9000 receiver also accepts this format. 7 = 3/1 23 (7). Used with older Sescoa or other receivers that send a 2300 Hz acknowledgment tone. The Model 9000 receiver also accepts this format. 7 = 3/1 23 (7). Used with older Sescoa or other receivers that send a 2300 Hz acknowledgment tone. The Model 9000 receiver also accepts this format. 7 = 3/1 23 (7). Used with older Sescoa or other receivers that send a 2300 Hz acknowledgment tone. The Model 9000 receiver also accepts this format. 7 = 3/1 23 (7). Used with older Sescoa or other receivers that send a 2300 Hz acknowledgment tone. The Model 9000 receiver also accepts this format. Note: This SIA formats are the preferred formats for the 5204, and are required if using the computer downloading feature.		

	Displayed In	formation		
Step #	On the 5230	On the Built-In Touchpad	Description	
Step 69	CIC #1	[Blank]	Carrier Identification Code is the prefix that needs to be dialed before a phone number to access a particular long distance carrier. Use special characters to add pauses, #, *, and "look for second dial tone" characters into the phone number. Enter A (SHIFT 1) for a pause. Enter B (SHIFT 2) for an asterisk (*). Enter C (SHIFT 3) for a number symbol (#). Enter D (SHIFT 4) to look for 2nd dial tone.	
Step 70	PHONE #1	1	Enter a phone # up to 16 digits long. Enter A (SHIFT 1) for a pause. Enter B (SHIFT 2) for an asterisk (*). Enter C (SHIFT 3) for a number symbol (#). Enter D (SHIFT 4) to look for 2nd dial tone. Any unassigned spaces will automatically be programmed as F. outside number 9A1D8885551212 area code pause 2nd phone number dial tone	
Step 71	ACCOUNT #2	2	Enter the account # for central station phone #2 (6 digits; leading zeros if shorter). Factory default account number = 205204	
Step 72	ATTEMPTS #2	З	See Step 67 (3-5)	
Step 73	FORMAT #2	5	See Step 68 (0-7)	
Step 74	CIC #2	[Blank]	See Step 69	
Step 75	PHONE #2	2	Enter a phone # up to 16 digits long (see Step 70). Enter A (SHIFT 1) for a pause. Enter B (SHIFT 2) for an asterisk (*). Enter C (SHIFT 3) for a number symbol (#). Enter D (SHIFT 4) to look for 2nd dial tone. Any unassigned spaces will automatically be programmed as F.	
Step 76	COMPUTER ACCOUNT	٥٥	Program the account number (6 digits; leading zeros if shorter) you want to use when uploading or downloading. When calling the computer, the 5204 uses phone line 1 and makes only one attempt.	
Step 77	COMPUTER CIC	[Blank]	See Step 69	

	Displayed Information		Description	
Step #	On the 5230 On the Built-In Touchpad			
Step 78	COMPUTER PHONE	З	See Step 70	
Step 79	TEST TIME	٥	Enter the time of day time using the 24-hour military format (4 digits) that you wish to send a TEST report to the central station. Note that any events that have not been restored will be sent along with the TEST report. The TEST will report first followed by the unrestored events.	
Step 80	CURRENT TIME	[Blank]	Enter the current time using the 24-hour military format (4 digits). It is advisable to check the time every few months, and reset it if necessary. Note: When using the Model 5541 downloading software, use the View Status menu to set the time. Refer to the Model 5541 Downloading Software manual (P/N 150497) for more information.	

	Option Number and Corresponding Word				
00	Blank	31	EQUIPMENT	62	REMOTE
01	# 1	32	EXIT	63	REST
02	#2	33	1ST	64	ROOF
03	#3	34	1ST FLR	65	ROOM
04	#4	35	FLOOR	66	2ND
05	# 5	36	4TH	67	2ND FLR
06	# 6	37	4TH FLR	68	SENSOR
07	#7	38	FRONT	69	SHIPPING
08	# 8	39	GARAGE	70	SHOP
09	# 9	40	GATE	71	SMOKE
10	# 10	41	GENERATOR	72	SOUTH
11	AC	42	GROUND	73	SPRINKLER
12	ACCOUNT	43	HALL	74	STAGE
13	ALARM	44	HEAT	75	STAIRWELL
14	AREA	45	HVAC	76	STORAGE
15	ATTIC	46	LAB	77	SUPER
16	BACK	47	LEVEL	78	TAMPER
17	BANQUET	48	LOADING	79	3RD
18	BASEMENT	49	LOBBY	80	3RD FLR
19	BOILER	50	LOCATION	81	TROUBLE
20	CELL	51	LOFT	82	UPPER
21	CENTER	52	LOWER	83	VALVE
22	COMPUTER	53	LUNCHROOM	84	VAULT
23	CONFERENCE	54	MECHANICAL	85	WAREHOUSE
24	DOCK	55	MEZZANINE	86	WATERFLOW
25	DOOR	56	NORTH	87	WEST
26	DUCT	57	OFFICE	88	WHSE
27	EAST	58	OPERATOR'S	89	ZONE
28	ELECTRICAL	59	PHONE	29	ELEVATOR
60	PULL	30	ENTRY	61	PULL STATION

Table 8-2: Programmable Word List

Section 9 Troubleshooting

When the system is configured properly, the voltage readings on the input and output terminals should be the same as those shown in the terminal description in Section 5.6. Zone Troubleshooting (Section 9.2.2) can help you with this.

The factory-programmed value for code 0 is 5204; for code 1, it is 1111.

If the Model 9000 receiver prints HELP after downloading has taken place, you need to upgrade the Model 9000 software. Refer to Section 3 for contacting Silent Knight for an upgrade.

9.1 Silencing Notification Devices

External silence input is designed to silence notification devices only to protect the system from being disabled without an audible tone. To activate external silence input, press the SILENCE key or use the external silence keyswitch (Section 6.5). The on-board beeper and local annunciators (Model 5230) continue to sound, warning of the need for additional service. After a service person has checked the system, the alarm can be reset. Alarm reset will unlatch the zone and silence the on-board beeper and remote annunciators. The alarm memory can then be cleared and the system returned to normal.

To silence the on-board beeper and local annunciators, if necessary, refer to Section 7.1.3.

9.2 Earth Ground Fault Troubleshooting

9.2.1 P3 and P4 Fault

A P3 trouble indicates that the control has detected a short between circuit ground and earth ground. A P4 trouble indicates a short between one of the control power terminals and earth ground.

To determine the location of the short, remove field wiring circuits until the control returns to normal operation. When the circuit that caused the trouble is found, use an ohmmeter to measure the resistance between each wire in the circuit and earth ground mounting screw (located in the bottom left corner of the PC board). The resistance should be higher than 220K ohms.

9.2.2 Accu-Zone, Troubleshooting (Mode 25)

Accu-Zone, Troubleshooting allows you to determine the voltage on any zone input and most system supervisory input (including AC power and battery power) without using a voltmeter. All alarms and troubles are disabled while you are using mode 25 so that you can trip sensors, adjust wiring, and so on. This mode is intended to be used with the Model 5230 Remote Annunciator as some features of zone troubleshooting are not available with the built-in touchpad.

Using Accu-Zone Troubleshooting

1. Press 2 5 ENTER + code 0 to enter mode 25.

The first line of the display will show the channel (input) number (Table 9-1 shows which channel corresponds to each input). The second line will show a high, low, and present voltage measurement value.



These values, or "step numbers", are not actual voltage readings. Use the information following these steps and Table 9-1 to determine the values acceptable for each input, and to calculate the actual voltage reading.

- 2. The high and low values allow you to momentarily trip a zone, then come back to the touchpad and see the result. This is also useful in locating intermittent connections. If you are using the built-in touchpad, only the present voltage appears on the display.
- 3. Press ENTER to advance to the next channel.
- 4. Press STEP to skip to a different channel. For example, if you were on channel 5 and you pressed STEP, the LCD would appear as follows:

ENTER CHANNEL #5 BELL 1: 23 23 23

Enter the channel number and press ENTER.

5. Press STEP STEP CLEAR CLEAR to exit mode 25.

The values can range from 0 to 31 and are shown in the 12-volt and 24-volt columns of Table 9-1. To arrive at the voltage reading for each terminal, use the following formula:

Actual voltage = present step number X n

Where n = Volts per Step value (in table)

For example, if the Zone 3 reading is " $3:0601\underline{06}$ ", The actual voltage would be calculated as $06 \times .15 = .9 \text{ V}.$

	Zone Input						
Channel		Terminal	12-Vo	lt Panel	24-Vo	Volts per	
		Number	Valid Range	Typical Value	Valid Range	Typical Value	Step
1-4	Zones 1-4 (Class A (style D))	18, 20, 23, 25	4-19	5	6-24	10	0.016
5-6	Notification circuits 1-2	7-8, 9-10	10-15	12	20-28	24	0.08
7	Accessory power (+AUX)	1	-	11	-	21	1.27
8	Touchpad power (KEY+)	3	-	11	-	11	1.27
9	Smoke power (+SMK)	19, 21, 24	-	11	-	21	1.27
10	AC power	AC Connector	≥ 22	28	≥ 22	30	.90*
*For AC powe	er measured on 120 V wiring	, the Volts per S	tep is approx	ximately 3.60.			
11	Battery power **	Cables	<u>></u> 9	9	<u>></u> 17	21	1.30
** Before taki	ng a battery reading, discon	nect AC power a	and put the pa	anel into an alar	m condition.		
12	Earth ground	Earth mounting screw	≤ 14 ≥14	8 20	$\leq 14 \\ \geq 14$	8 20	0.160

Table 9-1: Mode 25 Voltage Calculations

Special Notes

- 1. On Bell and Supervisory, a value below the minimum or above the maximum will cause a trouble.
- 2. On earth ground, the present voltage should be constantly changing between 8 and 20. If the reading does not cover this range, it could mean that the earth ground is shorting to something on the PC board.
- 3. Any voltage too high to measure will have the value 31. This is normal only for class B (style B) alarms.

9.3 Troubleshooting and System Messages

Table 9-2 shows the messages that may appear on the Model 5230 touchpad display and the codes that may appear on the 5204 built-in touchpad display.

For troubleshooting, you can connect a 5230 temporarily if it is not part of the installation (see Section 5.9).

5230 Message	Description or Action	Built-in Touchpad Display or LEDs
(Channel data)	A zone test is being conducted.	-5 SET MODE LED on
(Individual option names)	The 5204 is in the Step Programming mode.	- 7 SET MODE LED on
(Cycling system messages)	Smoke power is being tested.	- 1
(Cycling system messages)	The alarm memory has just been cleared.	-2
(Cycling system messages)	Dialer is being reset.	- 3
ALARM ZONE#	The alarm memory is being displayed.	- 5
FIRE #0 FIRE DRILL	A fire drill or system test is in progress.	SET MODE LED on ALARM LED on
ALARM ZONE 1-4	An alarm condition exists in the indicated zone.	J-4 ALARM LED on
DISABLE: ZONE # ZONE DESCRIPTION	Disabled (shunted or bypassed) zone. (The zone descriptions appear only if selected as a programming option.)	ы - <i>ь</i> ч
DIALER TROUBLE	The optional dialer has tried the programmed number of times (TOTAL ATTEMPTS) and has not been able to communicate with the central station (DIALER FAILED condition). Dialer failed. The communicator has failed to report.	dF
DL	Data lost. Communicator has lost data it was trying to transmit to the central station.	dL
WALK TEST	A walk test is being conducted. The top line of the 5230 display may also show the zone number in the trouble condition.	-2 SET MODE LED on
PRE-ALARM ZONE X ZONE LOCATION	An alarm condition exists in the indicated zone, but will not sound and report alarm until pre-alarm time has elapsed. During pre-alarm time, pressing RESET ALARM ENTER (code 1 or 0) will prevent sounding and reporting.	

Table 9-2: System Messages and Codes

Table 9-2: System	Messages	and	Codes
-------------------	----------	-----	-------

5230 Message	Description or Action	Built-in Touchpad Display or LEDs
REPORTING	An event is being reported to the central station.	REPORT LED flashes
SILENCED	A trouble condition exists and the annunciator has been turned off.	SILENCE LED on
SMOKE ZONE #	Smoke verification time, zone 1-4.	d1 · d4
SPRINKLER # ZONE DESCRIPTION	Sprinkler supervisory alarm.	[1] - [4]
SYSTEM NORMAL	No trouble, alarm, or other condition exists.	
TIME?	The 5204 is in the Set Time mode.	- 9 SET MODE LED on
LOW AC	AC power has been lost. Check connection to AC power source.	RC TROUBLE LED on AC/DC flashes
LOW BATTERY	Battery power has been lost, or polarity has been reversed. Measure the battery voltage and replace the battery or reverse polarity if necessary.	dl TROUBLE LED on AC/DC flashes
TROUBLE BELL #1/#2	A trouble condition exists on the indicated notification device.	R1-R2
TROUBLE GROUND	An earth to circuit ground fault condition exists. Use mode 25 to locate and correct the condition. Earth ground shorted to power. Use mode 25 to locate and correct the problem.	P3 TROUBLE LED on P4 TROUBLE LED on
TROUBLE REMOTE #	One or more of the Model 5230 annunciators is in trouble.	<i>F1 - F3</i> TROUBLE LED on
TROUBLE TROUBLE ZONE #	Sprinkler supervisory trouble.	TROUBLE LED on
TROUBLE LINE 1	A trouble condition exists on phone line 1.	L 1 TROUBLE LED on
TROUBLE LINE 2	A trouble condition exists on phone line 2.	L2 TROUBLE LED on
TROUBLE ZONE #1-#4	A trouble condition exists in the indicated zone. Refer to	7_U
(Zone location on second line of LCD.)	Section 9.2.2 to find and correct the trouble condition.	TROUBLE LED on
TRY AGAIN	A keystroke error has been made. Press CLEAR) and enter the correct keystrokes.	
TROUBLE COM 1	Unable to report manual or auto test using line 1.	L1 TROUBLE LED on.
TROUBLE COM 2	Unable to report manual or auto test using line 2.	L2
UNDEFINED #	Alarm condition on undefined type zone.	Ul - U2 ALARM LED on.

When the 5230 annunciator is powered up, it will show its ID number (1, 2, or 3) followed by the cycle of messages describing conditions that are currently in effect.

While the 5204 is communicating with the central station, the LCD will show the following message:

(Cycling Message) REPORTING

If two or more zones are in alarm, the top line will cycle through the status messages for these zones.

When the transmission is completed, the 5230 annunciator memory is reset (cleared) and the annunciator ID number is displayed.

10.1 Power Loss Reporting

The 5204 monitors both AC and battery power. The AC report delay time can be programmed in the range of 6-15 hours. An AC power-loss condition will cause the audible trouble condition to sound and the green LED to flash on and off. The 5205 will report the trouble to the central station (after the programmed delay time).

For NFPA Central Station Fire Alarm systems, AC report time must be 6-12 hours.

For NFPA Remote Station Protected Fire Alarm systems, AC report time must 15 hours.

10.2 Reporting Formats

The Model 5204 can transmit information in the following formats. The type of format you select is determined by the type of receiver used at the central station (see Section 8.4.38).

SIA8 and SIA20	Security Industry Association standard.
Silent Knight 3/1	Old format, transmits a 3-digit account number and a 1-digit alarm code. Transmissions are acknowledged at 1400 Hz.
Sescoa 3/1	Old format, transmits a 3-digit account number and a 1-digit alarm code. Transmissions are acknowledged at 2300 Hz.
Silent Knight 4 + 2	Silent Knight tone burst format, transmits a 4-digit account number and a 2-digit alarm code.
Silent Knight FSK	High-speed, single-round format for use with the Model 9000 and older receivers. Transmits a 4-digit account number and 2-digit alarm code.

The tables in the following sections show the digits that are transmitted for each event reported by the 5205 dialer, and the message printed out by the Model 9000 receiver at the central station.

Caution

Some formats do not distinguish between certain types of reports, such as between waterflow and fire alarms, or between supervisory and trouble reports. The central station must keep records of how the various zones are programmed at each account, so they can determine what condition is being reported for a particular zone.

10.2.1 SIA Format Printed Messages

In the SIA8 and SIA20 formats, the 5204 transmits the English description shown in the first column of Table 10-1. At the central station, the 9000 receiver prints the English message shown in the second column.

5204 Dialer Condition	9000 English Language Printout				
Alarm 1-4	"FIRE" ALARM 1-4 *				
Alarm Restore 1-4	"FIRE" ALARM RESTORE 1-4 *				
Disable 1-4	"FIRE" SHUNTED 1-4 *				
Disable Restore 1-4	"FIRE" SHUNT RESTORE 1-4 *				
Trouble 1-4	"FIRE" TROUBLE 1-4 *				
Trouble Restore 1-4	"FIRE" TROUBLE RESTORE 1-4 *				
AC Lost	A.C. TROUBLE 0				
AC Restore	A.C. RESTORE 0				
Battery Trouble	TROUBLE 09				
Battery Restore	LOW BATTERY 0				
Manual Test	BATTERY RESTORE 0				
Automatic Test	AUTO TEST 0				
Fire Drill	MANUAL TEST				
Downloading succeeded*	PROGRAMMING PASS 0 **				
Downloading failed*	PROGRAMMING FAIL 0 **				
Phone Line #1 Trouble	PHONE LINE TROUBLE 1				
Phone Line #2 Trouble	PHONE LINE TROUBLE 2				
Annunciator # 1-3 Trouble	EXPANSION TROUBLE 17-19				
Bell #1 Trouble	EXPANSION TROUBLE 32				
Bell #2 Trouble	EXPANSION TROUBLE 33				
Earth Ground Trouble, Ground	EXPANSION TROUBLE 38				
Earth Ground Trouble, Power	EXPANSION TROUBLE 39				
Data Lost	DATA LOST 0				
Phone Line #1 Restore	PHONE LINE RESTORE 1				
Phone Line #2 Restore	PHONE LINE RESTORE 2				
Annunciator # 1-3 Restore	EXPANSION RESTORE 17-19				
Bell #1 Restore	EXPANSION RESTORE 32				
Bell #2 Restore	EXPANSION RESTORE 33				
Earth Ground Restore, Ground	EXPANSION RESTORE 38				
Earth Ground Restore, Power	EXPANSION RESTORE 39				

Table 10-1: 9000 Printout for SIA Format

* All zones can be programmed as FIRE, WATERFLOW, UNDEFINED, or SPRINKLER. Fire is used as an example in the 9000 printout column above; the actual word printed will be whatever zone type has been programmed (see Steps 6-9 in Section 8).

**The 9000 must have the Model 9307 software package, Revision 900501 or later, to print the PROGRAMMING PASS and PROGRAMMING FAIL messages.

10.2.2 Silent Knight 3/1 and Sescoa 3/1 Formats

These formats transmit a 3-digit account number and a single-digit alarm code. These two formats greatly limit the amount of information that can be reported. To avoid confusion at the central station, standard alarm digits should be chosen. During programming, you select which alarm digits will be reported for different events. You can choose not to report restores or not to use zone numbers that might be duplicated by a supervisory transmission.

The second column in Table 10-1 shows the programming step number in which each digit is programmed. The third column indicates the 9000 prints only the digit (X) that has been programmed for that event - NOT an English message.

Note: When using the 3/1 formats, many of the reporting capabilities are lost because of the limited number of codes that can be sent.

5204 Dialer Condition	Programming Step Number	9000 Printout		
Alarm 1-4	55	CODE X = ALARM		
Alarm Restore 1-4	58	CODE X = RESTORE		
Disable 1-4	57	CODE X = DISABLE		
Disable Restore 1-4	58	CODE X = RESTORE		
Trouble 1-4	56	CODE X = TROUBLE		
Trouble Restore 1-4	58	CODE X = RESTORE		
AC Lost	56	CODE X = TROUBLE		
AC Restore	58	CODE X = RESTORE		
Battery Trouble	56	CODE X = TROUBLE		
Battery Restore	58	CODE X = RESTORE		
Manual Test	59	CODE X = TEST		
Automatic Test	59	CODE X = TEST		
Fire Test	59	CODE X = TEST		
Downloading succeeded*	59	CODE X = TEST		
Downloading failed*	59	CODE X = TEST		
Phone Line #1 Trouble	56	CODE X = TROUBLE		
Phone Line #2 Trouble	56	CODE X = TROUBLE		
Annunciator # 1-3 Trouble	56	CODE X = TROUBLE		
Bell #1 Trouble	56	CODE X = TROUBLE		
Bell #2 Trouble	56	CODE X = TROUBLE		
Earth Ground Trouble	56	CODE X = TROUBLE		
Data Lost	59	CODE X = TEST		
Phone Line #1 Restore	58	CODE X = RESTORE		
Phone Line #2 Restore	58	CODE X = RESTORE		
Annunciator # 1-3 Restore	58	CODE X = RESTORE		
Earth Ground Restore	58	CODE X = RESTORE		
Bell #1 Restore	58	CODE X = RESTORE		
Bell #2 Restore	58	CODE X = RESTORE		

Table 10-2: Printout for Silent Knight and Sescoa 3/1 Formats

* A test report might indicate that the 5204 has been downloading.

10.2.3 Silent Knight FSK and 4+2 Formats

The Silent Knight FSK and 4 + 2 formats transmit a 4-digit account number and a 2-digit alarm code. When an event is reported in either of these two formats, the dialer transmits the two digits shown in the second column. The 9000 can be programmed to print either the two digits or the English message shown in the third column of Table 10-3. If you are using the 9032 line card, FSK2 and BFSK are the only formats that will report in English.

Caution

Do NOT use the FSK or 4 + 2 formats when using the uploading or downloading functions. These formats do not distinguish between different types of tests (see Table 10-3).

5204 Dialer Condition	Digits Transmitted	9000 English Language Printout (9002 Line Card)		
Alarm 1-4	01-04	ALARM 01 - ALARM 04		
Alarm Restore 1-4	21-24	ALARM RESTORE 01 - 04		
Disable 1-4	51-54	SHUNT 11 - 14		
Disable Restore 1-4	21-24	ALARM RESTORE 11 - 14		
Trouble 1-4	61-64	TROUBLE 01 - TROUBLE 04		
Trouble Restore 1-4	71-74	RESTORE 01 - RESTORE 04		
AC Lost	60	A.C. TROUBLE		
AC Restore	70	A.C. RESTORE		
Battery Trouble	69	LOW BATTERY		
Battery Restore	79	BATTERY RESTORE		
Manual Test	30	TEST		
Automatic Test	30	TEST		
Fire Test	30	TEST		
Downloading succeeded*	30	TEST		
Downloading failed*	30	TEST		
Phone Line #1 Trouble	31	PHONE LINE TROUBLE 01		
Phone Line #2 Trouble	32	PHONE LINE TROUBLE 02		
Annunciator # 1-3 Trouble	33	EXPANSION TROUBLE		
Bell #1 Trouble	33	EXPANSION TROUBLE		
Bell #2 Trouble	33	EXPANSION TROUBLE		
Earth Ground Trouble	33	EXPANSION TROUBLE		
Data Lost	39	DATA LOST		
Phone Line #1 Restore	35	PHONE LINE RESTORE 01		
Phone Line #2 Restore	36	PHONE LINE RESTORE 02		
Annunciator # 1-3 Restore	37	EXPANSION TROUBLE		
Earth Ground Restore	37	EXPANSION TROUBLE		
Bell #1 Restore	37	EXPANSION TROUBLE		
Bell #2 Restore	37	EXPANSION TROUBLE		

Table 10-3: 9000 Printout for FSK & SK 4+2 Codes

* A test report might indicate that the 5204 has been downloading.

10.2.4 Radionics BFSK Format

The Radionics BFSK format transmits the event descriptions shown in the first column of Table 10-4. The second column shows the English message that the 9000 receiver prints at the central station.

5204 Dialer Condition	9000 English Language Printout				
Alarm 1-4	ALARM 01 - ALARM 04				
Alarm Restore 1-4	RESTORE 01 - RESTORE 04				
Disable 1-4	TROUBLE 01 - TROUBLE 04 / FORCE ARMED				
Disable Restore 1-4	RESTORE 01 - RESTORE 04				
Trouble 1-4	TROUBLE 01 - 04				
Trouble Restore 1-4	RESTORE 01 - 04				
AC Lost	TROUBLE 00				
AC Restore	RESTORE 00				
Battery Trouble	TROUBLE 09				
Battery Restore	RESTORE 09				
Manual Test	RESTORE 0E				
Automatic Test	RESTORE 0E				
Fire Test	RESTORE 0E				
Downloading succeeded*	RESTORE 0F				
Downloading failed*	TROUBLE 0F				
Phone Line #1 Trouble	TROUBLE 0B				
Phone Line #2 Trouble	TROUBLE 0C				
Annunciator # 1-3 Trouble	TROUBLE 0D				
Bell #1 Trouble	TROUBLE 0D				
Bell #2 Trouble	TROUBLE 0D				
Earth Ground Trouble	TROUBLE 0D				
Data Lost	TROUBLE 0E				
Phone Line #1 Restore	RESTORE 0B				
Phone Line #2 Restore	RESTORE 0C				
Annunciator # 1-3 Restore	RESTORE 0D				
Earth Ground Restore	RESTORE 0D				
Bell #1 Restore	RESTORE 0D				
Bell #2 Restore	RESTORE 0D				

Table 10-4: 9000 Printout for Radionics BFSK Format

* In Radionics BFSK format, the 9000 does not print alarm type, just the words ALARM, TROUBLE, etc.

Appendix A Programming Quick Reference

The quick reference table in this appendix briefly describes all of the available programming options and lists the factory-programmed default values that come with the Model 5204 panel. The 5204 can be programmed from a Model 5230 Remote Annunciator or with the Model 5541 downloading software.

For each programming option, the quick reference table identifies the menu or step to use for whichever programming method you choose. The table also provides an area for you to write in your selections before you begin programming. For more detailed information about a particular programming option, page references are included.

Option	Default	Your Selection	5230 Step #	5541 Menu	Page #	
Annunciator Operation						
Maximum Number of Supervised 5230s	0		3	В	8-7	
Current Time Displays on LCD	Blank		76	Through Request Status	8-26	
Bell and Other Audible Signaling Options	•	•				
External Silence Delay	Yes		2	В	8-7	
Bell #1 Silence	No		34	D	8-16	
Bell #2 Silence	No		39	D	8-17	
Bell #1 Output Pattern (zones 1-4)	Steady		35-38	D	8-16	
Bell #2 Output Pattern (zones 1-4)	Steady		40-43	D	8-17	
Sound Smoke Delay	No		24	В	8-14	
Device Enables - Enabling devices used with t	he 5204.	•				
Enable Computer	Yes		47	Е	8-18	
Enable Dialer	No Dialer		46	Е	8-18	
Dialer and Reporting Options						
Dialer Type	No Dialer		46	Е	8-18	
Central Station Phone Number 1	Blank		67	Е	8-23	
Central Station Phone Number 2	Blank		72	Е	8-25	
Central Station Account Number 1	005204		65	Е	8-22	
Central Station Account Number 2	005204		69	Е	8-24	
Events Reported to Central Station:						
3/1 Alarm code	1		55	Е	8-21	
3/1 Trouble code	8		56	Е	8-21	
3/1 Sprinkler code	2		57	E	8-21	

Table A-1: Quick Reference Table

Option	Default	Your Selection	5230 Step #	5541 Menu	Page #
3/1 Disable code	5		58	Е	8-21
3/1 Restore code	7		59	Е	8-21
3/1 Test code	9		60	Е	8-21
Computer Account Number (For uploading or downloading.)	005204		73	E	8-25
Computer Phone Number (For uploading or downloading.)	Blank		74	E	8-25
Fail Attempts for Account #1 (Number of attempts before local annunication of DIALER TROUBLE condition.)	3		66	Е	8-23
Fail Attempts for Account #2 (Number of attempts before local annunication of DIALER TROUBLE condition.)	3		70	Е	8-24
Number of Rings to Activate Downloading	10		54	Е	8-20
Ground Start	No		48	Е	8-19
Low AC Delay (hrs)	6		53	Е	8-20
Telephone Number To Report:					
Alarm #1 1st	Yes		61	E	8-21
Trouble #1 1st	Yes		62	Е	8-22
Disabled Zones (Shunted Zones) #1 1st	Yes		63	Е	8-22
Test #1 1st	Yes		64	Е	8-22
Must Report #1	Yes		51	E	8-19
Must Report #2	No		52	E	8-20
Touch-tone Line 1	No		49	E	8-19
Touch-tone Line 2	No		50	Е	8-19
Dialer Format #1	SIA 8		67	Е	8-23
Dialer Format #2	SIA 8		71	Е	8-25
General Purpose Relay #2 Options					
Relay #2 Usage	Alarm		33	D	8-16
Secret Code Options	L	L			
Code #0 (Installer's code) (Access to all functions.)	5204		44	D	8-18
Code #1 (Main user's code) (Access to all functions except programming.)	1111		45	D	8-18
System Options - Options that control overall function of 5204.					
24 Volt Smoke Power	24V=Yes 12V=No		1	В	8-7
Current Time Displays on LCD	Blank		76	Through Request Status	8-26
System (Troubleshooting) Messages	See Sect. 9.3		N/A	N/A	9-4
Test Options	1	1	•		

Table A-1: Quick Reference Table

Option	Default	Your Selection	5230 Step #	5541 Menu	Page #
Test Time	01:30		75	E	8-26
Report Tests to Telephone #1 1st	Yes		64	В	8-22
Timer Options					
Smoke Reset Time	2 sec.		4	В	8-8
Test Time	01:30		75	E	8-26
Pre-Alarm Time	5 sec.		18	В	8-14
Smoke Verification Time	6 sec.		23	В	8-14
Zone Options					
Zone Type (zones 1-4)	Fire		6-9	В	8-8
Zone Response Speed (zones 1-4)	.34 sec.		29-32	В	8-15
Pre-Alarm Delay (zones 1-4)	No		19-22	В	8-14
Pre-Alarm Time (zones 1-4)	5		18	В	8-14
Smoke Verification Delay	6		23	В	8-14
Smoke Verify (zones 1-4)	No		25-28	В	8-15
Allow Zone Disable	Yes		5	В	8-8
Zone Location Description (zones 1-4)	Blank		10-13	С	8-9
Cross Alarm (zones 1-4)	No		14-17	В	8-10

Table A-1: Quick Reference Table